



# Article Challenges and Strategies for Achieving High Energy Efficiency in Building Districts

Xiaoting Chen <sup>1,†</sup><sup>(D)</sup>, Behrang Vand <sup>2,†</sup><sup>(D)</sup> and Simone Baldi <sup>1,\*</sup><sup>(D)</sup>

- School of Mathematics, Frontiers Science Center for Mobile Information Communication and Security, Southeast University, Nanjing 211189, China
- <sup>2</sup> School of Computing, Engineering & the Built Environment, Edinburgh Napier University, Edinburgh EH10 5DT, UK; b.vand@napier.ac.uk
- \* Correspondence: simonebaldi@seu.edu.cn
- <sup>+</sup> These authors contributed equally to this work.

Abstract: Achieving climate neutrality requires reducing energy consumption and CO<sub>2</sub> emissions in the building sector, which has prompted increasing attention towards nearly zero energy, zero energy, and positive energy communities of buildings; there is a need to determine how individual buildings up to communities of buildings can become more energy efficient. This study addresses the scientific problem of optimizing energy efficiency strategies in building areas and identifies gaps in existing theories related to passive design strategies, active energy systems, and renewable energy integration. This study delineates boundaries at the building and community scales to examine the challenges of attaining energy efficiency goals and to emphasize the intricate processes of selecting, integrating, and optimizing energy systems in buildings. The four boundaries describe: (B1) energy flows through the building envelope; (B2) energy flows through heating, ventilation, air conditioning and energy systems; (B3) energy flows through individual buildings; (B4) energy flows through a community of buildings. Current theories often treat these elements in isolation, and significant gaps exist in interdisciplinary integration, scalable frameworks, and the consideration of behavioral and socioeconomic factors. Achieving nearly zero energy, zero energy, and positive energy communities requires seamless integration of renewable energy sources, energy storage systems, and energy management systems. The proposed boundaries B1–B4 can help not only in analyzing the various challenges for achieving high energy efficiency in building communities but also in defining and evaluating these communities and establishing fair methods for energy distribution within them. The results demonstrate that these boundaries provide a comprehensive framework for energy-efficient designs, constructions, and operational practices across multiple buildings, ensuring equitable energy distribution and optimized performance. In addition, the definition of boundaries as B1-B4 contributes to providing an interface for energy-efficient designs, constructions and operational practices across multiple buildings.

**Keywords:** energy-efficient building types (NEB, nZEB, ZEB); building energy systems; sustainable building design; energy efficiency strategies; renewable energy integration; positive energy communities (PEC)

# 1. Introduction

The building sectors, encompassing commercial, residential, and public buildings, represent a substantial percentage of global energy consumption, accounting for approximately 30% to 40% of the total [1]. Moreover, these sectors make a substantial contribution to greenhouse gas emissions, accounting for approximately 40% to 50% of the total emissions [2]. Given the crucial role that buildings play in the context of climate change and global warming [3], it is imperative that investigations are conducted to ensure that buildings are as environmentally friendly as possible throughout all stages of their lifecycle.



Citation: Chen, X.; Vand, B.; Baldi, S. Challenges and Strategies for Achieving High Energy Efficiency in Building Districts. *Buildings* **2024**, *14*, 1839. https://doi.org/10.3390/ buildings14061839

Academic Editors: Felipe Feijoo and Daniele Groppi

Received: 11 May 2024 Revised: 13 June 2024 Accepted: 15 June 2024 Published: 17 June 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Among these stages, the operational stage, which encompasses the operation of building envelopes, energy systems, renewable energy (RE) technologies, and electric vehicles (EVs), among others, holds particular significance [4].

Building physics plays a significant role in determining the energy performance of a building during its operational phase [5]. For instance, the building envelope, which comprises external walls, roofs, base floors, windows, and doors [6], functions as a storage mechanism for energy, akin to a battery. Through the utilization of advanced materials and techniques, the building envelope effectively isolates indoor conditions from the transient outdoor conditions, thereby minimizing heat losses in colder months and heat gains during warmer seasons [7]. By optimizing the various components of building physics, it is possible to not only reduce the energy demands of a building but also to assist in maximizing energy storage within the building [6,8,9].

Beginning at least with the two energy crises of the 1970s, a significant focus of research in the field of building energy has been placed on improving energy efficiency [10,11]. One effective means of achieving this goal is through the optimization of building energy systems, such as heating, ventilation, air conditioning (HVAC) systems. This approach has been demonstrated to have a positive impact on reducing the environmental footprint of buildings. For example, Lee et al. [12] examined boiler control schemes in office buildings and found that a sequential control algorithm with a boiler grading ratio of 3:7 resulted in maximized efficiency and a reduction of approximately 7% in gas usage. Baldi et al. [13] evaluated a parametric cognitive adaptive optimization for building energy management systems (EMS) in office buildings, reporting substantial energy efficiency by managing the HVAC units. Wang et al. [14] investigated a school building equipped with a displacement ventilation unit and heat recovery facility, reporting that increasing the heat recovery efficiency reduces the energy demand for ventilation and cooling. Liu et al. [15] identified that higher coefficient of performance water-cooled chillers could reduce cooling system energy consumption compared to air-cooled chillers. These studies, among others, have underscored the critical role of building energy systems as a cornerstone in the pursuit of energy efficiency [10–16].

As the availability of fossil fuel energy diminishes, and awareness of the environmental harms associated with such resources increases, RE sources have become the fastest-growing energy sources for buildings and are expected to play a significant role in shaping the future of the world [17-19]. Viable RE systems for buildings typically include photovoltaic (PV) panels, wind turbines, solar thermal energy, combined heat and power systems (CHPs), and ground source heat pumps (GSHPs) [18]. These RE sources can be exploited for on-site or distributed energy supply to buildings [20,21] and can be utilized to provide electricity, heating, and cooling [22]. The utilization of RE sources is integral to the achievement of green buildings and green communities, as evidenced by various studies [19,23–25]. A specific examination conducted by Ordóñez et al. [19] evaluated the energy potential of PV systems in Andalusia, Spain, taking into account factors such as the building type, the available roof space, and the PV panel arrangement. The research found that if all residential buildings were equipped with PV systems, the solar PV production capacity could reach 9.73 GW/year, which would suffice to meet 78.89% of energy demand. Additionally, GSHPs have been acknowledged as a cost-effective solution to address environmental concerns within the building sector, with the potential to decrease heating and cooling requirements by 20%–40% and 30%–50%, respectively [26]. It is noteworthy that RE systems may allow for surplus electricity generation, which can be sold to the grid or other buildings, ultimately leading to the realization of zero energy or even positive energy buildings (PEBs) and communities.

It is more efficient, from an energy perspective, to utilize the surplus electricity locally within buildings as opposed to selling it to the grid. EVs can play a significant role in this. The global adoption of EVs is rapidly increasing, and EV charging can significantly contribute to the overall electricity demand of buildings [27,28]. Furthermore, the batteries of EVs have the potential to serve as distributed energy storage, mitigating fluctuations

from RE sources [29] and offering peak load reductions [3]. There are various ways in which EVs can be employed, such as load shifting, vehicle-to-grid [30], and vehicle-tobuilding interactions [31]. Zhou et al. [32], for instance, developed an advanced batteryprotected energy control strategy and mathematical model to describe real-time battery degradation for multi-directional energy interactions between vehicles, buildings, and the grid. Barone et al. [33] evaluated the potential of a building-to-vehicle-to-building scheme in Europe, which stores renewable electricity in the batteries of EVs and distributes it among multiple buildings, and found that this scheme could reduce electricity demand from the grid by up to 71%. Huda et al. [34] studied the potential use of EVs in the Indonesian power system and found that they could lead to a 2.8% (coal) and 8.8% (natural gas) reduction in peak hour supply. EVs offer a bridge between energy efficiency and RE [35].

In recent years, there has been a growing interest in the concept of zero-energy buildings (ZEBs) as a means of reducing energy consumption and improving energy efficiency in the built environment. The origins of this concept can be traced back to the early 2000s when the first proposals for zero-energy buildings were put forth [36]. In the following years, the concept of ZEBs gained traction, with a marked increase in research and development in this area beginning around 2006 [37,38]. ZEBs are designed to surpass traditional buildings in terms of building physics, efficiency of building energy systems, and integration of RE technologies. Since the inception of this concept [10], there has been a continued evolution in the development of ZEBs, leading to the emergence of related concepts such as nearly zero energy buildings (nZEBs) and net zero energy buildings [39]. Looking to the future, the next step in the evolution of ZEBs is the concept of PEBs [40], which holds the potential to not only meet their own energy needs but also to distribute surplus energy to the grid, EVs, and potentially neighboring buildings.

In recent years, there has been a growing body of literature that has begun to consider the potential benefits of utilizing RE resources for reducing primary energy consumption, energy sharing, and reducing reliance on the utility grid at the community, neighborhood, or district level, rather than at the individual building level [41,42]. This shift in focus is driven by the recognition that it is more efficient and impactful to achieve energy targets on a community scale, as opposed to an individual building scale. One of the first definitions of a zero energy community (ZEC) was published by the National RE Laboratory in 2009 [43]. In 2014, a research project presented a definition of a Nearly Zero Energy Neighborhood [44]. More recently, the term Zero Emission Neighborhood was coined by the Research Centre on Zero Emission Neighbourhoods and Smart Cities in 2018 [45].

In recent years, several European Union projects have been oriented towards the concept of a positive energy community (PEC) with the aim of efficiently and flexibly harnessing the RE generation and storage potential of communities and minimizing the impact on the centralized grid [46]. In 2018, the European Strategic Energy Technology (SET) Plan Action 3.2 Smart Cities and Communities Implementation defined the concept of positive energy district (PED) [47]. PECs and PEDs are regarded as effective solutions to facilitate energy transition and reduce CO<sub>2</sub> emissions in cities [48,49]. The goal of PECs and PEDs is to realize a balance between energy efficiency, energy flexibility, and energy production in order to achieve climate neutrality and energy surpluses [50]. These concepts encourage the adoption of RE and offer an enabling environment for sustainable lifestyles for residents [51,52].

In this study, a wide range of literature has been reviewed to provide a comprehensive understanding of the current state of research in energy efficiency strategies for buildings. This extensive review has identified significant scientific deficiencies and engineering challenges, particularly in integrating passive and active energy systems, renewable energy sources, and behavioral factors. The primary scientific problem addressed in this research is optimizing strategies for achieving high energy efficiency in building areas to meet sustainability goals and reduce carbon emissions. While existing theories provide valuable insights into passive design strategies (e.g., building orientation, natural ventilation, thermal mass), active energy systems (e.g., HVAC optimization, energy storage, smart grids), and renewable energy integration (e.g., solar panels, wind turbines), significant gaps remain. These gaps include the need for interdisciplinary integration that combines these elements effectively, the development of scalable and adaptable frameworks that can be applied from individual buildings to communities, and the inclusion of behavioral and socioeconomic factors that influence energy efficiency. Addressing these gaps highlights the need for more interdisciplinary approaches and scalable solutions. The current research aims to address these issues by proposing a comprehensive framework that integrates these elements effectively, thus advancing the field of energy-efficient building design and operation. This approach will enable the creation of comprehensive and effective strategies, ensuring practical application and maximum impact on energy efficiency and sustainability in building areas.

Despite these concepts being defined some years ago, it is well-recognized in the field that there is a lack of systematic frameworks to define the boundaries between negative energy, nearly zero energy, zero energy, and positive energy scenarios, both at the building and at the community scale. This study presents a comprehensive overview of two distinct scales of buildings and communities based on a historical research roadmap, to establish the boundaries of building scale and community scale, and to examine the role of buildings as modifiers of energy, climate, and cost. The motivation of this paper is to present the corresponding challenges and barriers through a deeper understanding of the current state of research on buildings and communities and to provide guidance for further research. For instance, the imperative pursuit of nearly zero energy communities (nZECs), ZECs, and PECs is paramount in curbing energy consumption and enhancing energy efficiency, while taking into account local climate conditions for optimal outcomes. As an approach, the seamless integration of energy efficiency into communities of building design and construction is pivotal in attaining nZECs, ZECs, and PECs. However, challenges persist in coordinating energy-efficient practices and establishing equitable methods for energy distribution within communities. Furthermore, cutting-edge technologies, such as realtime monitoring and data analytics, can bolster the efficacy of community-level energy efficiency strategies.

The structure of this manuscript is organized as follows: Section 2 outlines the methodology employed in the study. Section 3 provides a comprehensive analysis of the role of buildings as modifiers of energy, climate, and cost, supported by the examination of case studies. Section 4 discusses the challenges and future prospects associated with different boundary layers. Finally, Section 5 offers a discussion, and Section 6 presents a summary of the study.

## 2. Methodology

The examination of building performance has evolved from an initial focus on building envelope improvement to an emphasis on overall building energy performance. More recently, this focus has expanded to encompass the performance of communities of buildings. This examination encompasses various aspects such as the building envelope, the efficiency of energy systems and the integration of RE technologies. An important aspect of this examination is the alignment of different energy systems, including the relationship between energy systems and buildings, buildings and utility grids, and buildings within a community of buildings. In order to achieve higher or optimal performance, it is necessary to establish different boundary layers, identify the energy flows that cross these boundaries, and examine the research challenges associated with each layer of boundaries.

The process of establishing boundaries is executed in a two-fold approach. The initial step involves the conceptualization of three layers of boundaries for individual buildings. The second step involves the examination of the extension from the individual level to the community level and the establishment of boundaries for building communities.

For the purpose of inspecting an individual building, Figure 1 depicts the design in accordance with the first law of thermodynamics [53]. Supposing the building incorporates RE systems, three layers of boundaries encompass:

- The primary focus is directed towards the first layer boundary (B1), which encompasses the building envelope and internal heat gains, encompassing lighting, appliances, and occupants. This emphasis on optimizing the building envelope as a priority is driven by its pivotal role in effectively reducing the power rating of HVAC systems. The required energy to operate the building envelope ( $Q_1$ ) can be supplied by the RE systems or purchased from the utility grid. In some cases, e.g., with some type of facades, the building envelope may interact with the HVAC. In addition to mitigating losses to the external environment, the building envelope plays a crucial role in fulfilling the occupants' visual and acoustic comfort criteria meaning that  $C_1 \simeq P_1$ .
- The second layer boundary (B2) is based on HVAC systems and EVs. The required energy to operate the HVAC systems and EVs ( $Q_2$ ) can be supplied by the RE systems or purchased from the utility grid. Likewise, the HVAC systems and EVs allocate a portion of their energy consumption towards the building envelope, while the remaining portion is dedicated to fulfilling the occupants' thermal comfort and indoor air quality needs meaning  $C_2 \simeq P_2$ . It is presupposed that the building has been designed with the intent of meeting the occupants' comfort requirements.
- The third layer boundary (B3) is considered for the RE systems. This layer accounts for the generation of a surplus of energy within the building (*Sell*), which can be monetarily compensated through its sale to the utility grid.



**Figure 1.** Boundaries of individual buildings.  $Q_1$ : Energy consumed by the building envelope;  $W_1$ : Energy to satisfy visual and acoustic comfort requirements ( $P_1$ ), and losses of the building envelope;  $\Delta U_1$ : Energy stored in B1.  $Q_2$ : Energy consumed by HVAC and EVs;  $W_2$ : Energy to satisfy thermal comfort and indoor air quality requirements ( $P_2$ ) and to the building envelope;  $\Delta U_2$ : Energy stored in B2.  $Q_3$ : Solar and wind energy to RE systems;  $W_3$ : Energy produced by RE;  $\Delta U_3$ : Energy stored in B3.  $Buy_1$ : Energy bought from the grid for the building envelope;  $Buy_2$ : Energy bought from the grid for HVAC and EVs; *Sell*: Energy sold to the grid.  $C_1$ : Visual and acoustic comfort requirements of occupants;  $C_2$ : Thermal comfort and indoor air quality requirements of occupants.

According to the magnitude relationship between  $Buy_{sum} = Buy_1 + Buy_2$  and  $Sell_{sum} = Sell$ , buildings can be classified into four categories. The classification of buildings as negative energy buildings (NEBs), nZEBs, ZEBs, and PEBs is predicated upon the level of yearly energy consumption in relation to energy generation. Table 1 quantitatively demonstrates this relationship by categorizing buildings based on their energy consumption levels relative to their energy generation capacity as follows: greater than, almost equal to, equal to, and less than.

**Table 1.** Categories of buildings/communities according to the magnitude relationship between of *Buy*<sub>sum</sub> and *Sell*<sub>sum</sub> (yearly analysis).

Comparison	Term	Definition
Sell <sub>sum</sub> < Buy <sub>sum</sub>	NEBs / NECs	The level of energy consumption exceeds the energy generation
$Sell_{sum} \approx Buy_{sum}$	nZEBs / nZECs	The energy required from the grid is close to zero or very low, as most of the energy demand is supplied by RE sources
Sell <sub>sum</sub> = Buy <sub>sum</sub>	ZEBs / ZECs	RE balance the energy demand
Sell <sub>sum</sub> > Buy <sub>sum</sub>	PEBs / PECs	The energy generation exceeds the energy required for building's operation

For a community of buildings, the design is presented in Figure 2, taking two buildings as an example. In addition to the three-layer boundary of each building, a fourth-layer boundary (B4) is established through the inclusion of the energy flows between buildings. In addition, similar to the individual building scenario, depending on the magnitude relationship between  $Buy_{sum} = Buy_{1a} + Buy_{2a} + Buy_{1b} + Buy_{2b}$  and  $Sell_{sum} = Sell_a + Sell_b$ , communities of buildings also can be classified into four categories: negative energy communities (NECs), nZECs, ZECs, and PECs, refer to Table 1. It is worth mentioning that communities/districts/neighborhoods are used as synonyms in this work.



**Figure 2.** Boundaries of a community of buildings. Most of the energy flows have a similar meaning as in Figure 1. The new energy flows to be taken into account are the energy flows between buildings.

## 3. Buildings as a Modifier of Energy, Climate, and Cost

This section examines the impact of buildings on energy consumption, climate change, and economic costs. As depicted in Figure 3, the analysis is structured into two primary subsections: 3.1 focuses on individual buildings, including nZEBs, ZEBs, and PEBs, while 3.2 addresses communities of buildings, encompassing nearly nZECs, ZECs, and PECs. Critical components such as building envelopes (B1), energy systems (B2), renewable energy technologies (B3), and energy sharing mechanisms (B4) are also explored to elucidate their contributions to enhancing building performance and sustainability.



Figure 3. Conceptual framework for energy efficiency in buildings and communities.

#### 3.1. Individual Buildings

The high energy demands and low energy performance of the vast majority of existing buildings, commonly referred to as NEBs, have been well-documented in literature [54,55]. A significant proportion of global energy consumption is attributed to these structures [56]. Consequently, the enhancement of energy efficiency in existing buildings has been identified as a crucial strategy for reducing energy consumption and carbon emissions on a global scale. In recent years, various countries have devoted substantial resources towards achieving this goal, with building retrofits specifically garnering widespread attention across the globe [57–59]. For example, Wang et al. [60] offered an HVAC system retrofit scheme for an existing office building in Tianjin, resulting in an overall energy efficiency improvement of 71.2%. In general, standard retrofit actions can reduce the total energy consumption, although the building will still remain an NEB in most cases. In case of major retrofit actions, it is possible to upgrade a NEB to a nZEB, as illustrated below.

## 3.1.1. Nearly Zero Energy Buildings (nZEBs)

The concept of nZEBs has gained significant global attention, as evidenced by its inclusion in the 2010 European Union Directive [61], which defines it as "a building that has a very high energy performance. The nearly zero or very low amount of energy required should be covered to a very significant extent by energy from renewable sources, including energy from renewable sources produced on-site or nearby". Furthermore, the European Union Directive stipulates that all buildings occupied or owned by public authorities after 31 December 2018, should adhere to nZEB standards, and that all new buildings should conform to these standards by 31 December 2020 [61]. China, Japan, and the US are among the countries at the forefront of promoting nZEB goals. These nations have demonstrated a strong commitment to sustainable and energy-efficient building practices. For example, their initiatives include setting nZEB targets, implementing policies and regulations, and adopting strategies to drive nZEB adoption in the building sector. Their efforts encompass comprehensive analyses of China's nZEB policies, regulations, and strategies, as well as insights into Japan's policy frameworks, technological advancements, case studies, challenges, and lessons learned [62,63]. Given that energy conservation is often more feasible than its generation, it is widely recognized that reducing energy demand tends to be more cost-effective and efficient than generating energy on-site through RE sources [64]. This principle holds particularly true in the context of nZEBs.

The implementation of nZEBs has been extensively studied. Amani et al. [64] conducted a case study on a residential building in Tehran, Iran, which was retrofitted to nZEB standards through the addition of four layers of internal insulation, resulting in a 74% reduction in energy consumption. Ferrari et al. [65] retrofitted an office building on the campus of Politecnico di Milano, Italy with an improved thermal insulation envelope, replacement and upgrade of HVAC systems, renovation and installation of advanced lighting controls, and the incorporation of PV systems, ultimately reducing primary energy demand and green gas emissions by up to 40%. Hamdy et al. [66] presented a simulation optimization method and found that a single-family house equipped with PV and GSHP in a cold climate in Finland could achieve nZEB standards with a primary energy consumption of 70 kWh/m<sup>2</sup>year. Zhao et al. [67] renovated the facade of a residential building in Dezhou, China, and optimized the angle of PV panels to achieve nZEB standards. This solution lowered the building's energy costs by 14.1%, increased PV power generation by 24.13%, and reduced CO<sub>2</sub> emissions by 4306.0 kg CO<sub>2</sub>eq/year. Visa et al. [68] transformed a solar house in Romania, which was already a low-energy building with a geothermal system and solar energy converters, into a nZEB with less than 3% of the energy demand covered by the grid.

## 3.1.2. Zero Energy Buildings (ZEBs)

The concept of zero energy and emissions buildings emerged in the early 2000s, as a means of drastically reducing energy needs and carbon emissions in buildings through the realization of efficiency gains and the balance of energy needs being met through RE sources [69,70]. Significantly, the recast Energy Performance of Building Directive has introduced a notable shift in the standard for new buildings. Specifically, it mandates zero emissions buildings as the requirement from 2028 for all newly constructed buildings. Moreover, for new buildings occupied, operated, or owned by public authorities, the transition to zero emissions is expected even earlier, starting in 2026. This update marks a crucial milestone in the pursuit of sustainable building practices and signifies a concerted effort to align with environmental objectives [71]. It is expected that ZEBs play a further significant role in the advancement of smart cities by contributing to the achievement of goals related to energy efficiency, conservation, and RE generation [72].

In response to the growing concern for energy efficiency and sustainability, a number of countries have begun to investigate the potential of zero energy and emissions buildings. Kwan et al. [73] designed an affordable ZEB based on a typical house in Brisbane, Queensland by optimizing the building envelope, incorporating energy-efficient lighting and appliances, and adopting solar hot water collectors and a 5kW crystalline silicon PV panel system. Rey-Hernández et al. [74] studied the "LUCIA" building, a zero-energy building that reduced non-renewable resource depletion by 31% and significantly reduced the building's impact on climate change and soil fertility loss by using RE sources (biomass, geothermal and PV systems). Nord et al. [75] designed an integrated RE system consisting of flat plate solar thermal collectors (STCs) in combination with a GSHP and an exhaust air heat pump proposed for a single-family demo dwelling located in Larvik, Norway, as a demonstration zero-emission building. Lou et al. [76] studied a school building in Hong Kong, which achieved ZEB standard through the adoption of high-performance building envelopes, energy-efficient air-conditioning systems and lighting fixtures, and building-integrated PV (BIPV). Lindberg et al. [77] investigated cost-optimal solutions for the energy system design in a ZEB, using the case study of a German multi-family house equipped with GSHP, CHP, gas boiler, air source heat pump, STC and PV. The results indicated that a larger PV size was required to reach the zero energy balance.

## 3.1.3. Positive Energy Buildings (PEBs)

The advancement towards the enhancement of building efficiency has led to the emergence of the concept of PEBs. PEBs are characterized by their ability to generate energy in excess of that required for the operation of the building, for example through the utilization of RE systems to power EVs charging [39,78,79]. Cole et al. [80] emphasized that the aim of PEBs is not only to generate surplus energy but also to pay attention to the effective utilization and distribution of these excess resources. This surplus energy can be sold to the grid outside the building [39,81].

In recent years, PEBs have gained significant popularity across various nations. Ai et al. [82] conducted a study on a hybrid PV/wind system installed on Waglan Island, Hong Kong, and proposed a matching design approach that optimally determined the best configuration to fulfill the load demand at a minimal cost. Kolokotsa et al. [83] presented the design and energy technologies of a PEB school in Thessaloniki, Greece, and analyzed the energy performance which demonstrated a reduction of nearly 68% in energy demand, while significantly improving indoor thermal comfort. Bennani et al. [84] examined a high-rise office building in Morocco and applied energy efficiency measures to the baseline building, revealing a 56% reduction in energy consumption. Additionally, the integration of PV double glazing into the envelope reduced energy consumption by 15%, converting it to a PEB. Casini et al. [85] presented the results of the ReStart4Smart project, which achieved a PEB that was more than twice as energy efficient as Dubai's best practice through design choices of building form, building envelope, home systems, appliances, and EMSs in the geographical and cultural context of the Dubai region. Zomer et al. [86] presented the evolution of the energy balance and performance of all PV and BIPV systems at the Solar Energy Research Laboratory in Florianopolis, Brazil, and found that the PV system provides 148% of the building energy demand, thus it could be considered not only a ZEB but also a PEB.

### 3.1.4. nZEBs, ZEBs and PEBs: Case Studies

The classification of nZEBs, ZEBs and PEBs is presented in Table 2, which summarizes a variety of case studies from the literature. Each case study is examined in detail and analyzed in accordance with its unique characteristics. As countries around the world strive to meet low carbon and energy efficiency targets, nZEBs, ZEBs, and PEBs are being developed in a wide range of countries and regions. Given the variation in climatic, infrastructural, and other conditions among countries and regions, the building envelope methods (B1), building energy systems (B2) and RE technologies (B3) employed in the design of nZEBs, ZEBs, and PEBs are site-specific, resulting in variations in building performance.

The cases presented in Table 2 are drawn from various countries and regions, resulting in a diversity of weather conditions. The performance and level of achievement of each case are influenced by the specific building envelope design, building energy systems installed and RE technologies employed. As Table 2 illustrates, the effective coupling of the building envelope with RE technologies, and the appropriate selection of RE technologies, are critical factors in achieving zero energy targets, as previously noted in [87]. Additionally, the level of energy trading and energy demand are crucial considerations in the design and operation of nZEBs, ZEBs, and PEBs. These building types all exhibit lower levels of carbon emissions, thus contributing to the mitigation of the greenhouse effect and global warming, the reduction in environmental pollution, and the modulation of the climate. Hence, the level of carbon emissions is a key metric for evaluating the performance of these buildings. By fortifying the building envelope, increasing the efficiency of energy systems, and incorporating RE sources, energy demand can be reduced, allowing for the attainment of energy targets. As energy demand decreases, the associated costs also decrease.

Country/ Region	Category	Building Envelope (B1)	HVAC and EV (B2)	RE Technologies (B3)	Key Performances	References
Iran	nZEB	A four-layer insulation system formed by 12 different thicknesses of typical insulation materials	HVAC	No	Reduced energy consumption by 74% and minimized CO <sub>2</sub> emissions.	[64]
Italy	nZEB	A thermally broken aluminum frame and double-pane glass, add polystyrene panels on walls and on the roof	HVAC	PV	Reduced primary energy consumption by 5% and greenhouse gas emissions by 40%.	[65]
Finland	nZEB	Good tightness and insulation level	AHU, GSHP	PV	Achieved lower primary energy consumption and lower CO <sub>2</sub> emissions.	[66]
Shandong Province	nZEB	Prefabricated concrete wall panels, glass curtain walls	HVAC	PV	Reduced energy cost by 14.1% and CO <sub>2</sub> emissions by 4306 kg CO <sub>2</sub> eq/year, increased PV power generation by 24.13%.	[67]
Romania	nZEB	A light metallic structure covered with insulated panels outwards and plasterboard inwards, double-glazed windows, walls with 20 cm of cellular polystyrene thermal insulation	Ground source heat exchanger, heat pump	PV, STC	Reduced energy demand and CO <sub>2</sub> emissions, increased RE generation.	[68]
Queensland	ZEB	High insulation level for external walls and ceiling, white reflective roof, tinted reflective window glaze	HVAC	Solar hot water collectors, PV	Achieved energy savings by 66%.	[73]
Spain	ZEB	Thermal insulation of walls and apertures, double glazed argon-filled windows, materials of low environmental impact and life cycle, zigzag shaped facade, plastered internal insulated walls, green roofs	Ground heat exchanger, CHP, biomass boiler	PV	Reduced the depletion of non-RE by 31% and greenhouse gas emissions.	[74]
Norway	ZEB	N/A	GSHP, exhaust air heat pump	STCs, PV	Reduced electricity use noticeably.	[75]
Hong Kong	ZEB	High performance building envelope	HVAC	BIPV	PV installations can balance the energy use.	[76]
Germany	ZEB	U-values are setting according to the German passive building standard	GSHP, CHP, air source heat pump	STC, PV	Reduced energy demand and $CO_2$ emissions.	[77]
Hong Kong	PEB	N/A	HVAC	PV, wind generator, battery	Minimized the cost.	[82]
Greece	PEB	Extra insulation in the external walls and roofs as well as low U-values for windows, a solar chimney	HVAC	PV	Reduced energy demand by almost 68%.	[83]
Morocco	PEB	Double glazing, sun-shade	HVAC	BIPV	Reduced energy consumption by $71\%$ and $CO_2$ emissions.	[84]
Dubai	PEB	Thermal envelope performances better than codes, ventilated roof, cool colors, automated windows, responsive solar shading, green wall	HVAC, EV	BIPV, STC, battery	Reduced contruction costs and energy needs.	[85]
Brazil	PEB	The asymmetrical lateral facades, the use of sustainable materials, the different mixed materials on the exterior (bricks, metal, and glass)	N/A	BIPV	Reduced CO <sub>2</sub> emissions and energy consumptions. PV sysems supplied 148% of energy demands.	[86]

Table 2. Summ	arv of some ca	se studies for	individual	buildings.

Table 2 reveals that the use of insulation in the building envelope is the most commonly implemented solution, as it has been demonstrated to be cost-efficient and effective, as well as one of the simplest strategies for reducing building energy demand, as noted in [64,88]. Accordingly, energy goals are typically achieved by first reducing energy demand through the enhancement of the building envelope and the improvement of building energy systems, followed by the installation of RE technologies to meet any remaining energy demand. Furthermore, Table 2 illustrates that solar and wind energies are significant RE sources that can be employed to address climate change, as previously mentioned in [89]. However, wind power has relatively limited applications in individual building environments due to the constraints of urban space and the seasonal variability of wind, as discussed in [90]. In situations where space is not a constraint, such as in [82], the complementary nature of wind and solar energy can be fully exploited by selecting an appropriate combination

of wind turbine and PV array capabilities, thereby significantly reducing power outage periods. Therefore, optimizing building performance requires matching technologies to each building's specific needs, from retrofitting to advanced designs like nZEB or ZEB. Customizing solutions ensures maximum efficiency and sustainability across diverse building types.

## 3.2. Communities of Buildings

There is a systematic history of building performance improvement, which can be traced back to the development of concepts such as nZEBs and ZEBs. However, for a significant number of buildings, it is not realistic to achieve these targets due to their high energy demand [87,91]. In recent years, there has been a growing emphasis on achieving zero energy goals at the community level, due to the advantages offered by load diversity, the ability to control, divert and store energy produced from various sources and regions, increased efficiency, reduced energy consumption, and enhanced energy resilience [87,92–94]. A community of buildings refers to a group of buildings, which may include residential, commercial, or other types of buildings, situated in close proximity and sharing some common areas [95,96].

## 3.2.1. Nearly Zero Energy Communities of Buildings (nZECs)

The low number of buildings that adhere to the new regulations, such as nZEBs, in comparison to those that do not, necessitates the implementation of innovative strategies in order to achieve decarbonized communities. One potential approach is to incorporate various RE technologies in these buildings, thereby enabling the sharing of surplus electricity within the community. This can aid in achieving the goal of nearly zero energy consumption at a community level, as evidenced by the work of Amaral et al. who have updated the definition of nZEBs to include the concept of nearly zero energy districts [97].

Currently, the trend towards reducing energy consumption and emissions from buildings is shifting towards a macro-level approach, specifically focusing on communities of buildings, such as nZECs. This orientation towards macro-level energy management is rapidly gaining momentum in the field. As exemplars of such research and initiatives, Synnefa et al. [98] presented an integrated and cost-effective system geared towards the implementation of nearly zero energy settlements. This system comprises solutions for the building envelope, building energy production, and energy management, and was implemented in four demonstration projects located in the European Union-in Cyprus, France, Italy, and the United Kingdom—with varying climates and building types. The results of the study indicate that the transition from a nZEB to a nZEC resulted in a significant reduction in energy consumption to less than 20 kWh/ $m^2$  per year and a cost reduction of at least 16%. Ullah et al. [8] identified and analyzed 23 nZEBs as case studies. Through this analysis, they were able to discern various mitigation and control strategies employed by outdoor heat sources, building adaptation technologies, and RE technologies, with the aim of reducing energy consumption, mitigating greenhouse gas emissions and promoting RE generation and thermal energy production. In their study, Hachem-Vermette et al. [99] presented a case study of a mixed-use neighborhood in Calgary, Canada. The neighborhood comprised over 1000 residential units and approximately 8000 square meters of commercial, office, and educational buildings. The authors proposed the implementation of a PV system, which was designed to cover all south-facing roof areas to generate electricity for the district. Through energy simulations for different building types, it was determined that stand-alone and attached houses had the potential to be energy-positive for a given climate condition within the district, while other building types evaluated in the scenario, such as apartment buildings, offices, and supermarkets, were able to generate only a small fraction of their energy consumption. Sameti et al. [100] considered a net-zero energy district comprising seven residential and office buildings located in Switzerland and studied the impact of storage on the buildings' performance. The authors proposed the implementation of a CHP unit, a separate auxiliary boiler, and a thermal storage tank

to satisfy the heat requirement of each building. Additionally, a PV array, a CHP unit, and a battery bank were proposed to meet the electrical load. The authors proposed that heat and power be exchanged among the buildings in the neighborhood through the heat and power transmission networks. The results of the study indicated that the adoption of an optimal district energy system with energy storage can deliver considerable economic and environmental benefits when compared to conventional energy systems, stand-alone energy systems, and net zero energy without storage. Wills et al. [101] presented a modeling approach for evaluating community-scale retrofit design alternatives aimed at converting an existing community to net-zero energy and applied it to a community comprising 50 single-detached dwellings located in Toronto, Canada. The case study demonstrated that the achievement of net-zero retrofitting can result in a significant reduction in annual energy imports by 80% and a reduction in associated greenhouse gas emissions by 95% for the community. The authors also illustrated that buildings that do not achieve net zero at the individual building scale can be balanced at the community level through the utilization of adjacent buildings with more usable roof areas and/or lower energy consumption intensity. Huang et al. [93] developed and validated an open-source nZEC virtual testbed based on a real-world nZEC located in Florida, US. The virtual testbed offers great flexibility in model selection and integrated simulation optimization, enabling a more comprehensive assessment of integrated control from the building to the grid level. The transition towards nZECs emphasizes a holistic approach to energy management that efficiently merges the upgrading of existing buildings with the integration of advanced management and technologies. This strategy is particularly effective in leveraging the collective potential of both new and older buildings, significantly enhancing sustainability and energy savings across the community. By focusing on comprehensive community-level transformations rather than isolated updates, nZEC initiatives profoundly impact energy efficiency, cost reduction, and environmental sustainability, demonstrating the substantial benefits of coordinated energy management.

#### 3.2.2. Zero Energy Communities of Buildings (ZECs)

In 2009, the US Department of Energy established the definition of a ZEC as "an energy-efficient community in which the actual annual delivered energy on a source energy basis is less than or equal to the on-site renewable exported energy" [44]. This concept was further explored by Katipamula et al. [102] in 2010, who conducted a comparison of ZEC and individual ZEB and found that the economies of scale inherent in the community approach can help achieve equivalent overall energy performance at a lower cost. However, it should be noted that the implementation of ZECs also presents certain limitations, such as the complexity of planning, design, and monitoring, as well as the challenges of engaging stakeholders [44].

In the context of this research, numerous countries have been focusing on the design strategies employed in the creation of ZECs. Mavrigiannaki et al. [103] conducted an analysis of real data obtained from the first year of monitoring a zero-energy pilot community. The results of the performance analysis indicated that the community successfully achieved its targets for net regulated consumption, RE production, and cost. Furthermore, a positive balance was attained when considering the total energy consumption and PV production of the community for the first five months of monitoring, beginning with the start of summer. Overall, the community was able to achieve a positive energy balance for each year of energy demand. Fouad et al. [104] proposed an innovative sustainable ZEC comprising eight distinct building designs and 52 buildings, in order to simulate a realistic community. The ZEC featured well-insulated walls and roofs, as well as the incorporation of a green roof. The results demonstrated that all buildings within the ZEC met the criteria of the zero energy concept and fully satisfied the energy demand while generating excess energy through the utilization of RE sources. Shnapp et al. [105] presented a case study of a ZEC in Denmark, which was able to reduce energy consumption by 50% through the implementation of the following three steps: (1) optimization of district heating systems

with new radiators, ventilation systems, solar storage, and heat pumps, (2) insulation of building envelopes, improved windows, use of smart meters and reduction in district heating temperature set points, and (3) the remaining energy demand was met through the use of thermal solar, PV, and BIPV. González et al. [106] investigated a zero-energy neighborhood of 400 houses in the Netherlands. They proposed passive design measures and distributed power generation to achieve the zero energy target and demonstrated that passive design is a fundamental prerequisite for reducing building energy demand. De León et al. [107] provided a solution for achieving zero energy targets in a residential district under a tropical climate in Panama. They reduced energy consumption through the optimization of bioclimatic and energy strategies, achieving savings of 31%, with the remaining demand met through the generation of solar electricity. It can be noted that the progression from nZECs to ZECs represents the next phase in sustainable urban development, demanding enhanced technical solutions and improved building qualities. This transition involves a deeper integration of advanced energy technologies and updated standards for building performance, enabling communities to not only reduce but potentially negate net energy usage and emissions on a broader scale. This shift to ZECs requires technical upgrades and a holistic approach to community planning and management to achieve the ultimate goal of zero net energy.

### 3.2.3. Positive Energy Communities of Buildings (PECs)

Given the complexity of the calculation procedure necessary to achieve the goal of a ZEC and the difficulties inherent in attaining an exact zero, a positive target is often easier to calculate and implement, which in turn accelerates the realization of carbon neutrality [105]. As a result, research related to PECs has been proposed and developed in recent years, and the concept is gaining increasing attention due to the pressing energy and environmental crisis and the critical role of the building sector in emissions reduction [108–110]. Derkenbaeva et al. [108] provide a definition of PECs as "an energy-efficient and energy-flexible urban area or a group of connected buildings, which produces net zero greenhouse gas emissions and actively manages an annual local or regional surplus production of renewable energy". This definition highlights that PECs are primarily concerned with energy conservation, emission reduction, and sustainability, which are conducive to the development of smart cities [111]. However, it should be noted that PECs typically consist of a mix of existing buildings and new constructions.

The concept of PECs has been widely adopted globally since 2018 [3]. Barone et al. [112] developed an EMS that facilitates the transfer of energy from a RE building to two traditional buildings using three EVs as energy vectors. The EVs were charged and discharged based on a scheduled trip and the state of charge of the battery, resulting in the system achieving an energy positive of 8.6 MWh/year. Rehman et al. [89] proposed a centralized solar district heating network that is integrated with a renewable-based electricity network to meet the heating and electricity needs of 100 houses. This RE system was comprised of PV panels, wind turbines, and a stationary power storage system. The results of the study indicated that the system was able to reduce imported electricity by 2 kWh/m<sup>2</sup>/year when the storage system was taken into account. Brennenstuhl et al. [113] presented a Plus-Energy settlement consisting of 23 residential buildings that were equipped with decentralized heat pumps and PV systems in Germany. The study found that under ideal conditions, cost savings of up to 50% can be achieved at the building level by optimizing the self-consumption of PV electricity. Bambara et al. [114] examined the potential of replacing one stand-alone house with two houses of equal living space on the same parcel, with the new house being designed to be energy efficient in terms of HVAC, envelope and fitted with BIPV. The results revealed that the retrofitted scheme was 30% more energy efficient and that residential densification could significantly contribute to the transformation of existing communities into resilient PEDs. Gouveia et al. [115] conducted a study on the historic district of Alfama in Lisbon, Portugal, through building retrofits. The study found that the potential energy demand for heating and cooling was reduced by 84% and 19%, respectively. Additionally, integrating building PV technology on roofs and windows potentially generated energy of 60 GWh/year. The results of the study demonstrated the potential for energy retrofit measures and solar PV products in households. It can be summarized that the transition to PECs offers a more practical route to achieving substantial carbon neutrality, proving to be more cost-effective for property owners compared to nZECs and ZECs. PECs, which focus on net-zero greenhouse gas emissions and renewable energy surplus, are evolving as a sustainable urban development strategy that combines energy conservation with advanced EMSs to enhance the energy efficiency of both new and existing buildings.

# 3.2.4. nZECs, ZECs and PECs: Case Studies

The classification of nZECs, ZECs, and PECs is based on factors such as country/region, categorization, building envelope, building energy systems, renewable energy technologies, the scale of the community (number of buildings and total floor area), and key performances, as detailed in Table 3. This is similar to the evaluation of individual buildings in Table 2, but in contrast to the individual building scenario, the consideration of scale is crucial as existing cases of communities of buildings vary in scope and magnitude.

The cases examined in this study originate from various regions characterized by distinct meteorological conditions, as well as varying building envelope designs, building energy systems, RE technologies and scales of communities. As a result, their performance exhibits significant variation. It is crucial to recognize that each factor contributing to differences in community performance represents a significant challenge that must be addressed in order to achieve greater efficiency. For example, in the context of PECs developed under the auspices of European Union funding from H2020, a minimum floor space of 15,000 m<sup>2</sup> and the inclusion of at least three large buildings (new construction, renovation, or a combination of both) [116] are mandatory requirements. Consequently, the scale of the community is of paramount importance and must be considered when determining the optimal location for B4 of the community.

The results of the case studies presented in this research demonstrate that targets for nZECs, ZECs and PECs can be attained through the reduction in energy consumption, CO<sub>2</sub> emissions, investment costs, and/or the enhancement of RE production. Furthermore, it has been established that a more optimal interface between the building envelope and RE technologies, as well as a more seamless interaction amongst RE technologies, leads to a reduction in investment costs. Thus, the compatibility of the building envelope and RE technologies, as well as the compatibility of RE technologies, is of critical importance. Additionally, energy trading between buildings, which allows for buildings that are not ZEBs to obtain energy from PEBs within a community of buildings, is also essential.

The primary sources of RE currently remain solar and wind; however, they are inherently intermittent and uncontrollable, resulting in temporary mismatches between energy supply and demand [117]. Seasonal mismatches [118] are particularly prevalent, such as in regions with harsh climates, where a mismatch exists between the large-scale heating demands in winter and the abundance of solar resources in summer [119]. In certain countries, such as Spain, non-RE flows are only required to be compensated by energy exports within the same month [120], indicating that this variability will lead to strain on the grid and ultimately compromise grid stability [3], which has direct implications for the realization of nZECs, ZECs and PECs. Similar to seasonal mismatches, there are also hourly mismatches, where grid challenges such as overvoltage may emerge in the summer due to the large size of building-based PV installations and the concentration of buildings in a small area [121]. These mismatches can only be addressed through the incorporation of short or long-term energy storage in a community of buildings [101]. Additionally, a more advanced type of energy storage, such as using EV batteries as mobile energy storage devices to serve as power sources for buildings, can improve the resilience and flexibility of buildings, enabling peak shaving and load shifting, and reducing the power consumption of buildings during peak hours [112].

Country/Region	Category	Building Envelope (B1)	HVAC and EV (B2)	RE Technologies (B3)	Scale (B4)	Key Performances	References
Cyprus, France, Italy, UK	nZEC	Advanced insulating material based on the new generation of extruded polystyrene	CHP, solar air conditioning	Combined solar and wind system, PV	123 buildings, 323608 m <sup>2</sup> ; 48 dwellings, 1400 m <sup>2</sup> ; 4 villas, 30,000 m <sup>2</sup> ; 500 family homes, area N/A	France, Italy and the UK reduced investment cost by 16%, while Cyprus reduced 12.4%. Energy use $\leq 20$ kWh/m <sup>2</sup> . RE production $\geq 50$ kWh/m <sup>2</sup> .	[98]
Calgary, Canada	nZEC	Wall and roof insulation of 7 m <sup>2</sup> K/W and 10 m <sup>2</sup> K/W, respectively, triple glaze, low-e argon fill windows, airtight construction	HVAC	BIPV	1003 residential units including 180 houses and 27 apartment buildings, area N/A	Schools and houses reduced annual primary energy consumption and $CO_2$ emissions, and increased RE production to offset the negative energy balance of most of the remaining buildings.	[99]
Switzerland	nZEC	N/A	CHP, thermal storage tank	PV, battery	7 buildings, area N/A; heat and electricity exchange	Reduced $CO_2$ emissions by 60%. Lower cost.	[100]
Toronto, Canada	nZEC	Exterior wall insulation, ceiling insulation, foundation wall insulation, triple-glazed low-e argon filled windows	GSHP, heat pump	PV	50 single-detached dwellings, area N/A	Reduced energy demand by 69%. Reduced $CO_2$ emissions by 95%.	[101]
Florida	nZEC	Cool roofs, low-E glass and extra insulation	PV	2 retail stores, one commercial office, and two residences, 2776 m <sup>2</sup>	GSHP	After operating for a half years, the annual electricity generation was very close but slightly less than production.	[93]
Italy	ZEC	Extruded polystyrene insulation	HVAC	PV, battery	2 single-family houses, 2760 m <sup>2</sup>	Reduced investment cost by 24%. Energy conservation was 75.7% compared to a standard community. Achieved RE production of 50.03 kWh/m <sup>2</sup> /year.	[103]
Egypt	ZEC	Insulated walls and roofs, green roof, single glazing windows	HVAC	PV, wind turbines	52 buildings, area N/A	Reduced energy consumption by 57.6%. Reduced CO <sub>2</sub> emissions by 390 tons.	[104]
Denmark	ZEC	Insulation of building envelopes, improved windows	HVAC	STC, PV, BIPV	Avedøre Green City, area N/A	Reduced energy consumption by 50%.	[105]
The Netherlands	ZEC	High-transmission and low emissivity glazing, user-controllable window shades, insulation walls, floors and roofs, 30mm-thick plaster layer to the walls	CHP, EVs	PV, battery, STC collectors	400 houses, area N/A	Reduced the primary energy demand for heating by 50%. In terms of electricity it can be positive.	[106]

**Table 3.** Summary of some case studies for community of buildings.

Country/Region	Category	Building envelope (B1)	HVAC and EV (B2)	RE technologies (B3)	Scale (B4)	Key performances	References
Panama	ZEC	The inclusion of local shading, double glass, and insulation roofs	HVAC	BIPV	34 houses, 1870 m <sup>2</sup>	Reduced energy consumption by 31%. PVs could cover 100% of the demand.	[107]
Italy	PEC	N/A	HVAC, EV battery	BIPV	3 buildings and 3 EVs, area $N/A$	Electricity saving of about 11.4% and economic saving of 8.1%.	[112]
Finland	PEC	N/A	Heat pumps, borehole thermal energy storage, EVs	Wind turbines, PV, STCs, battery	100 houses, area N/A	Maximized the onsite-energy fraction and onsite-energy matching.	[89]
Germany	PEC	N/A	Heat pumps	PV, battery	23 residential buildings, area N/A	Achieved cost savings up to 50%.	[113]
Québec, Canada	PEC	High insulation, double glazed argon low-e windows, airtight construction	HVAC	BIPV	2 houses, 220.176 m <sup>2</sup>	Reduced energy consumption by 30%. Generated nearly 3 times more electricity than they consumed.	[114]
Portugal	PEC	Roof (thermal insulation with mineral wool), PVC window frames with standard double glass, walls through internal insulation	HVAC	BIPV	6004 dwellings, area N/A	Reduced energy demand by 84% and 19% for space heating and cooling.	[115]

## 4. Challenges, Future Outlooks and Recommendations

The energy performance of various building communities has been evaluated through measurement or simulation, revealing that a multitude of factors significantly impact performance. These effective factors present challenges from various perspectives and necessitate a thorough examination of each. This research endeavors to focus on individual boundaries in order to identify and address the challenges associated with increasing the energy performance of individual buildings or building communities, as well as the intended distribution of energy and costs within the community, including energy storage systems. The challenges primarily center around the following aspects, such as macro-level target identification; planning evaluation; calculation of energy demand and supply; enhancement of the building energy systems (e.g., RE); identification of mismatches between energy supply and demand due to RE sources; degradation of EV batteries; adjustments to EMSs. These challenges are classified and grouped according to the boundaries specified in Section 3, with an overview provided in Figure 4, as detailed below.



**Figure 4.** Framework illustrating the boundaries and key components for enhancing energy efficiency in buildings and communities.

#### 4.1. Boundary 1: Building Envelope

The literature review and subsequent analysis have delineated the primary technical and practical challenges as follows:

**Passive Design Strategies:** Building design can be optimized to reduce energy consumption, for example by using natural light, optimizing the orientation of the building, and using passive design strategies such as thermal mass. Using high-reflectance roofing materials such as cool roofing can reduce heat gain in the summer and lower cooling costs. Green roofs, walls, and facades can help to reduce energy consumption by providing insulation, shading, and cooling effects.

**Insulation Technology:** Proper insulation of the building envelope can help to reduce heat loss and gain. Increasing the insulation levels in the walls, roof, and floors can help to reduce heat loss in the winter and heat gain in the summer, which can lower the amount of energy needed to heat and cool the building. The reduction in heat loss in the winter can be achieved through the manipulation of infiltration levels using sealing gaps and cracks in the building envelope. Specifically, the sealing of openings around doors and windows, as well as in the walls, floors, and ceilings, can effectively prevent drafts and thus lower the overall heat loss in the winter.

**Energy-efficient Windows Technology:** Installing energy-efficient windows can help to reduce heat loss and gain. For example, replacing old, single-paned windows with energy-efficient, double-paned windows equipped with shading devices can significantly reduce heat loss in the winter and heat gain in the summer. Installing shading devices such as overhangs, awnings, or shading screens on the south- and west-facing windows can reduce heat gain in the summer and increase energy efficiency. Incorporating passive solar design elements such as south-facing windows and thermal mass can reduce the need for heating and cooling.

**Proper Ventilation Technology:** Proper ventilation systems that bring in fresh air and remove stale air can help to keep the building comfortable, which can reduce the need for heating and cooling.

**Envelope Component Matching Technology:** Optimization of energy demand within a building can be achieved through the strategic alignment of building envelope elements with the annual energy consumption patterns. This process referred to as "envelope component matching", involves the selection and configuration of building envelope systems and elements that correspond to the specific energy demands of the building throughout the year.

**Retrofitting Challenges:** The building envelope assumes paramount importance in shaping the energy demand of buildings. Consequently, it becomes evident that focusing on the physical attributes of buildings represents the primary means to achieve substantial reductions in energy demand. This involves establishing novel guidelines for new constructions and undertaking comprehensive retrofitting measures for existing buildings. Despite the growing prevalence of building retrofits, the unique physical characteristics and preservation principles associated with historic buildings, for example, present significant limitations, warranting further consensus on the criteria to be considered. Consequently, the domain of retrofitting necessitates further refinement and advancement.

## 4.2. Boundary 2: HVAC and Energy Systems

**Energy Systems Selection Challenges:** The selection of appropriate energy systems for buildings is a complex process that is influenced by a variety of factors, including climatic conditions and geographical location. Despite the acknowledged importance of energy systems in building design, the methodologies for determining the most suitable energy systems for a specific building are not well-established. It is common for buildings to incorporate a variety of different energy systems; however, the interactions between these systems, and how they can be optimized for improved energy efficiency performance, remains poorly understood. The lack of a cohesive approach to energy systems integration in building design is a significant obstacle to achieving higher levels of energy efficiency in buildings. Therefore, it is crucial to develop a systems in buildings to improve overall energy performance. Furthermore, the interconnections between different energy systems in a building should be well-defined, and the ways in which they can interact to achieve higher efficiency performance should be thoroughly investigated.

**HVAC Equipment Sizing Challenges:** The critical role of proper HVAC equipment sizing in promoting energy efficiency and reducing waste in buildings cannot be overstated. An HVAC system must be able to provide adequate heating and cooling to maintain comfortable indoor temperatures while avoiding oversizing, which can cause energy

waste and negatively impact indoor air quality and equipment reliability. Conversely, undersized systems struggle to maintain desired temperatures, leading to increased energy consumption and decreased comfort. The American Society of Heating, Refrigerating and Air Conditioning Engineers (ASHRAE) recommends sizing HVAC systems based on building loads and occupancy patterns, considering factors such as insulation, airtightness, solar gains, and ventilation needs [122]. This can be achieved through energy simulation models, manual calculation methods, and design guidelines provided by organizations like ASHRAE. In conclusion, proper HVAC equipment sizing is crucial for reducing energy waste and improving efficiency in buildings. It enables optimization of energy performance, lower operating costs, and enhanced indoor comfort.

**Optimal Systems and Envelope Integration Challenges:** The optimal combination of building energy systems and building envelope has been shown to significantly reduce energy consumption,  $CO_2$  emissions, and energy costs. Consequently, it is essential to carefully consider the unique characteristics of each building in order to effectively match the appropriate energy systems and achieve superior building energy performance.

**Zoning Challenges:** Zoning a building into temperature control areas provides targeted and efficient heating and cooling, leading to improved energy performance. The installation of separate heating and cooling systems for zones with higher internal heat loads or lower heating and cooling requirements allows for individually controlled heating and cooling, resulting in energy savings and improved thermal comfort. The US Department of Energy reports that zoning can save 30–40% of heating and cooling energy consumption [123]. This is supported by the ASHRAE Handbook of Fundamentals, which states that zoning improves thermal comfort, reduces energy consumption, and increases control flexibility [122]. Zoning is, therefore, a critical aspect of building design and operation for enhancing heating and cooling efficiency and reducing energy consumption.

**Energy Storage Systems Challenges:** The challenge of balancing energy supply and demand is a complex issue, particularly in light of the diverse infrastructure and environmental conditions present in different regions. In this context, energy storage systems have been identified as a crucial element in achieving a balance. While energy storage has the potential to mitigate seasonal mismatches, this necessitates the implementation of large-scale storage capacity. The current state of the industry necessitates the development of standard, cost-effective seasonal and long-term energy storage systems for heating and cooling, as well as electricity [124]. Furthermore, the rational design of energy storage systems is crucial for the achievement of optimal performance.

**Routine HVAC Maintenance Challenges:** Routine HVAC maintenance, sealing and insulating ductwork, and upgrading equipment are crucial to building performance and energy management. Neglect of maintenance leads to decreased performance, increased energy consumption, and poor air quality. Routine inspections, cleaning, and addressing wear and tear improve longevity, reduce breakdown risk, and ensure compliance with regulations, safety, and health standards. Sealing ductwork stops leaks, improves air circulation, and reduces energy loss. Insulating ductwork regulates air temperature, reduces heat transfer, and improves efficiency, potentially reducing energy use by up to 20%. Upgrades, such as high-efficiency air filters, programmable thermostats, and variable-speed fans, can result in energy savings, improved indoor air quality, and reduced carbon footprint. Regular inspections and maintenance of the ductwork and HVAC system are essential to enhance overall building performance and prevent energy loss.

**EV Integration Challenges:** The integration of EVs into buildings or communities has been gaining increasing attention due to the potential for utilizing surplus electricity generated from RE sources, thus reducing emissions [125]. However, the high frequency of charging and discharging associated with EVs has been identified as a major challenge, as it can lead to accelerated battery degradation and impede the integration of EVs into the grid. This constitutes a significant barrier to the wider adoption of EVs in this context.

**Indoor Air Quality Challenges:** In the context of designing new buildings or retrofitting existing ones, it is imperative to acknowledge and address the indoor air quality [126]

20 of 32

needs of building occupants, encompassing aspects such as thermal comfort, indoor air quality, and visual and acoustic comfort. To ensure precise pre-analysis, the utilization of energy performance simulation tools becomes indispensable, as they facilitate accurate energy calculations and enable comprehensive assessments of thermal comfort [127–129].

## 4.3. Boundary 3: Individual Buildings

The pursuit of nZEBs, ZEBs, and PEBs is a complex and multi-faceted endeavor, requiring the resolution of a multitude of technical, financial, behavioral challenges, consistent implementation, monitoring and verification, and interdisciplinary challenges as presented, for example, in [64,89,130–132]. The realization of nZEBs, ZEBs, and PEBs necessitate a comprehensive understanding of these challenges and the development of effective strategies to overcome them. The following represents a list of the various challenges.

**Technical Challenges:** The main technical challenges associated with the pursuit of nZEBs, ZEBs, and PEBs encompass (1) the design and construction of energy-efficient building systems; (2) the integration of RE sources; (3) the optimization of building performance, specifically in terms of energy demand, cost reduction, and emissions mitigation; (4) the technical standards and regulations. Overcoming these technical challenges requires a comprehensive and interdisciplinary approach, encompassing expertise from various fields such as architecture, engineering, and RE.

**Financial Challenges:** The pursuit of nZEBs, ZEBs, and PEBs presents substantial financial obstacles that must be overcome to achieve these highly energy-efficient buildings. These challenges encompass a variety of factors, including elevated initial construction expenses, restricted access to financing and investment capital, a lack of knowledge regarding the long-term cost savings and benefits of nZEBs, ZEBs, and PEBs, and a scarcity of incentives or subsidies for sustainable building practices. Additionally, the integration of EMSs focusing on the operational period is crucial to cover all financial aspects and enhance the economic feasibility of these buildings. Addressing these financial barriers requires concerted collaboration between governments, building practices and promoting their economic viability.

**Behavioral Challenges:** The pursuit of nZEBs, ZEBs, and PEBs requires the resolution of various behavioral challenges, which pertain to human attitudes and behaviors that must be addressed to achieve these highly energy-efficient buildings. These challenges may encompass resistance to change, lack of understanding or awareness of the benefits of energy-efficient building practices, as well as an absence of incentives or motivations for individuals and organizations to embrace sustainable building practices. Moreover, issues pertaining to building operations and maintenance such as insufficient training for building occupants and staff, and lack of accountability for energy usage may also pose a significant challenge. Overcoming these behavioral barriers necessitates a collaborative effort from governments, building owners, contractors, and educational institutions to promote sustainable building practices and educate building occupants and staff about the importance of energy efficiency.

**Consistent Implementation and Adherence Regulations and Standards:** The consistent implementation and adherence to regulations and standards in the pursuit of nZEBs, ZEBs, and PEBs poses significant challenges that must be overcome to achieve these highly energy-efficient buildings. These challenges encompass a lack of clarity and uniformity in standards, difficulties in ensuring compliance, and an absence of transparency and accountability in the building design, construction, and operations process. Furthermore, the compatibility of new technologies with existing standards and the absence of adequate technical expertise can hinder effective implementation. Addressing these challenges requires a collaborative effort from governments, industry professionals, and educational institutions to establish clear, consistent regulations and standards, promote transparency and accountability, and provide technical expertise and training to support their successful implementation.

Monitoring and Verification Challenges: The challenges of monitoring and verification in the pursuit of nZEBs, ZEBs, and PEBs represent a formidable impediment to the achievement of highly energy-efficient buildings. These challenges may encompass a dearth of accurate and trustworthy monitoring systems, obstacles in attaining comparable and uniform data, and the complexity of verifying the energy efficiency of buildings. Furthermore, there may be difficulties in integrating monitoring systems with existing infrastructure and a shortage of technical knowledge to effectively operate and analyze the collected data. Addressing these challenges necessitates collaboration between governments, building industry professionals, and educational institutions, in order to establish accurate monitoring systems, implement consistent and comparable data-gathering techniques and equipment (e.g., IoT devices), and provide technical proficiency to ensure effective monitoring and verification of energy performance in these buildings.

Interdisciplinary Challenges: The main challenges related to interdisciplinary cooperation in the endeavor to attain nZEBs, ZEBs, and PEBs encompass concerns that must be resolved to ensure the accomplishment of these energy-efficient buildings. These challenges may involve the absence of synchronization and collaboration between various specialists and stakeholders, including architects, engineers, building proprietors, and contractors. Additionally, there may exist a lack of correspondence between the aims and objectives of diverse groups, resulting in conflicting design and construction decisions. With the aim of overcoming these interdisciplinary challenges, a robust dedication to collaboration, effective communication and mutual comprehension of respective roles and goals, and a common objective of creating highly energy-efficient buildings are crucial.

Addressing these challenges requires a concerted effort from multiple sectors, including governments, building industry professionals, and educational institutions. A key factor in overcoming these challenges is clear and effective communication, which promotes a shared understanding of the goals and objectives involved in the creation of energy-efficient buildings. Additionally, there must be a strong commitment to sustainable building practices and a willingness to collaborate and work together towards the common goal of creating buildings that are not only energy-efficient but also environmentally responsible and sustainable in the long term.

## 4.4. Boundary 4: Communities of Buildings

The scholarly inquiry and subsequent practical innovations within the domain of building communities are characterized by their technological sophistication and the exigent nature of contemporary challenges, as evidenced by the comprehensive body of work accumulated up to the completion of this study. The pursuit of nZECs, ZECs, and PECs poses a complex set of challenges that must be addressed to achieve their highly energy-efficient objectives. Some of these challenges include coordination of energyefficient design, construction, and operational practices across multiple buildings, a lack of established and universally accepted standards and guidelines, difficulties in ensuring compliance with regulations, and insufficient transparency and accountability in the design, construction, and operational processes. The integration of new and advanced technologies with existing standards and regulations, as well as the availability of specialized technical expertise and knowledge for effective implementation, are also significant challenges in this pursuit. The integration of RE sources, energy storage systems, and EMSs [133] into the community-wide energy system presents additional difficulties. In order to overcome these obstacles, a collaborative effort is needed from government agencies, building industry professionals, and educational institutions to establish clear and consistent regulations and standards, enhance transparency and accountability, and provide the necessary technical training and expertise to support successful implementation.

**Definition and Boundary Challenges:** The definition and boundaries of nZECs, ZECs, and PECs remain a topic of research and debate. Uncertainty surrounds the inclusion of infrastructure energy demand and the classification of energy loads as controllable or uncontrollable. The geographical boundary of these communities is also debated, with proposals

ranging from virtual neighborhood boundaries to physical city or regional boundaries. It is important to establish a clear definition and methodology for evaluating communities, taking into account the complexities of different energy systems and communities. Advanced technologies, such as real-time monitoring and advanced data analytics, may aid in overcoming current evaluation limitations. Therefore, it is needed to establish a consistent framework for these communities' analysis.

**Community Scale Challenges:** The scale of a community of buildings, including the number and physical dimensions of each building, is crucial for determining energy performance. Energy demand is influenced by the quantity of buildings and their scale, as larger buildings require more energy for heating, cooling, and lighting. The feasibility of incorporating RE sources, energy storage systems, and EMS is impacted by the community scale. Larger communities may face greater challenges in implementing RE sources and require more sophisticated EMSs. Conversely, smaller communities may be able to adopt simpler EMSs and integrate RE sources more easily. Overall, community-scale plays a significant role in energy performance.

**Energy Management Challenges:** As previously discussed, the EMS plays a pivotal role in enhancing energy efficiency and curbing consumption by facilitating synchronized interactions between buildings and the electrical grid. The efficacy of EMS is significantly influenced by the scale and complexity of the building community it serves. In compact communities with a limited number of buildings, EMS deployment tends to be more straightforward, allowing for seamless integration at either the individual building or community-wide level. This smaller scale permits a direct correlation between the EMS control strategies and the underlying mathematical models governing the energy systems. Conversely, in expansive communities encompassing a multitude of structures, the deployment of EMS becomes inherently more complex. The necessity for distinct simulation and programming tools to manage energy systems introduces potential discrepancies and their mathematical representations. To address these challenges in larger communities, the adoption of cutting-edge EMS technologies becomes imperative.

Energy Efficiency Improvement Challenges: Improving energy efficiency at the community level is crucial for reducing energy consumption and addressing climate change. Strategies should be designed with consideration of local climate conditions, such as temperature, humidity, and wind patterns, to optimize results. Evidence supports the importance of local climate consideration in energy efficiency strategies. For instance, in warm regions with high solar radiation, reducing cooling demands through shading and ventilation is more effective, while in colder regions, reducing heating demands with insulation and efficient heating systems is key. It can be noted that a one-size-fits-all approach is ineffective and that strategies must be tailored to local conditions to achieve optimal results. Moreover, the significant role of local climate in determining energy consumption patterns and that strategies that take these conditions into account are more likely to be effective. In conclusion, the design and implementation of energy efficiency strategies at a community level must take into account local climate conditions to achieve optimal results. To effectively enhance energy efficiency at the community level, it is crucial to figure out the relationship between local climate variables and energy usage, forming strategies that are conventional to each area's unique conditions. Furthermore, integrating cutting-edge technologies such as real-time monitoring systems, sophisticated data analysis, and robust communication networks will amplify the formulation and execution of more successful and sustainable energy efficiency measures for communities.

**Integrating Energy Efficiency Challenges:** Integrating energy efficiency into community building design and construction is crucial for achieving nZECs, ZECs, and PECs. The density of the community and available roof and facade area for RE generation must be considered. Integrating city and building design into energy planning ensures effective energy solutions. However, the method of sharing remaining grid imports among community buildings can impact energy performance [134]. Currently, there is no universally accepted and equitable method for distributing energy within communities and to the grid. In conclusion, nZECs, ZECs, and PECs can be achieved when communities of buildings are designed and constructed with high levels of energy efficiency and sufficient spaces, for example, roof and facade area for RE generation, provided the density of the community is not too high. An integrated planning process, incorporating city and building design into energy planning, is critical for the success of these initiatives. Additionally, further research is needed to establish a fair and widely accepted method for energy distribution within communities of buildings.

**Energy Transmission Challenges:** Energy transmission between buildings in a community requires an internal network. Financial and technical challenges, as well as main grid impact, may arise. Mitigating these challenges requires careful design and implementation, considering technical and economic feasibility. Adopting advanced technologies, such as smart grid systems and distributed energy resources, can improve efficiency and reliability [135]. Integrating energy storage systems and using RE sources, such as solar and wind power, can enhance stability, resilience and overall energy performance, reducing reliance on the main grid and sharing within the community. In conclusion, the development of an effective internal energy network for communities of buildings requires a comprehensive approach that takes into account both the technical and economic feasibility, as well as the protection and reliability of sharing electricity in both the local and main grid. The use of advanced technologies, energy storage systems, and RE sources can help overcome some of the challenges associated with the development of an internal energy network, and ensure its success.

High Initial Investment Challenges: It is essential to consider long-term benefits; therefore, the design of sustainable buildings and communities faces a major challenge in the high initial investment required. A comprehensive and integrated approach is needed, taking into account the local climate, future climate trends, building type, energy costs, system operation, and techno-economics, to balance economic viability, environmental sustainability, and energy efficiency while ensuring comfort. Conflicting objectives may require multi-objective optimization models to identify trade-offs and find the optimal solution. High-efficiency building technologies and materials, such as insulation and air-tightness systems, and the integration of RE sources like solar and wind power can help reduce energy consumption and reliance on conventional energy sources, mitigating the challenge of the high initial investment. In conclusion, successful sustainable building cost-effectiveness, environmental impact, and energy efficiency. Considering multi-objective optimization and leveraging state-of-the-art technologies like RE sources and EMSs are key to addressing the challenges of high initial costs and achieving these objectives.

**Key Performance Indicators Challenges:** Evaluating the effectiveness of zero energy projects requires using key performance indicators to measure progress. Environmental and energy key performance indicators, such as RE produced, CO<sub>2</sub> reduction, and RE share increase, provide a neutral assessment. However, a lack of standard key performance indicators, calculation methods, and tools results in difficulty in quantifying energy use, emissions, and efficiency. The local resource potential and inherent characteristics of regions can affect project feasibility and success. A uniform assessment methodology referencing model assumptions and choices is needed to ensure comparability and transparency across projects and case studies. In order to overcome the lack of standard key performance indicators and assessment tools, there is a need for more research and collaboration between stakeholders to develop a consistent and comprehensive methodology for evaluating the performance of ZECs. Additionally, the use of advanced technologies, such as smart grid systems and distributed energy resources, can help to improve the accuracy and reliability of energy use and emissions data and provide a more comprehensive picture of the community's energy performance.

**Federated Decision-making Challenges:** Effective energy project planning demands a federated, iterative, and transparent approach to accommodate changes in technology,

policies, and processes within political, economic, cultural, and social contexts. A federated decision-making method integrates multiple stakeholder perspectives, leading to a comprehensive understanding of challenges and opportunities. Continuously monitoring and evaluating progress and adapting strategies accordingly is crucial for success. Accurate, real-time information is necessary for informed decision-making, achieved through robust monitoring and evaluation systems. It can be concluded that this approach enables decision-makers to effectively respond to evolving technologies, policies, and processes in different contexts and support the achievement of project goals.

Load Forecasting Challenges: Accurate forecasting of building energy demand and load is critical for optimized energy management in integrated systems. This precise load prediction is key to effectively managing and distributing energy resources. Current methods include building simulation software and data-driven technologies. However, simulation software may have low accuracy and data-driven methods lack in considering interrelationships between loads. Hence, advanced forecasting methods are needed to provide a comprehensive picture of future energy demand. This can be achieved by considering coupling relationships between loads and integrating real-time data from various sources with machine learning algorithms and big data analytics. The unique characteristics of the energy system and its loads must also be considered to improve forecasting accuracy. It can be noted that to ensure optimal scheduling of integrated energy systems, accurate and comprehensive energy demand forecasting is a necessity. Advanced energy demand forecasting methods need to be developed that take into account the complex relationships between multiple loads, and that use real-time data and machine learning algorithms to provide a more accurate and comprehensive picture of future energy demand. By doing so, energy systems can be managed more efficiently, leading to reduced energy costs, reduced greenhouse gas emissions, and improved energy security.

#### 5. Discussion

The building sector is responsible for a significant portion of global energy consumption and CO<sub>2</sub> emissions. Key measures to achieve green buildings and communities include optimizing building physics, building energy systems such as HVAC, and utilizing RE systems like PV panels and wind turbines. The surplus electricity generated by RE systems can be used for EV charging and as distributed energy storage, contributing to peak load reductions. Additionally, EVs offer multidirectional energy interactions between vehicles, buildings, and the grid, reducing electricity demand from the grid.

Building performance evaluation has evolved to encompass the energy efficiency of communities of buildings. Establishing different boundary layers is crucial for identifying energy flows that cross them and examining associated challenges. The three boundary layers for individual buildings are based on the building envelope, energy systems, EVs, and surplus energy. Buildings can be classified as negative energy, nearly zero energy, zero energy, or positive energy based on energy consumption and generation. For communities of buildings, a fourth-layer boundary is established to include energy flow exchange between buildings, and they can also be classified based on energy consumption and generation.

Efficient energy use in buildings is pivotal for reducing global energy consumption and curbing carbon emissions. NEBs exhibit high energy demands and poor performance, but standard retrofit measures can significantly lower their total energy consumption. nZEBs are increasingly prevalent, as reducing energy demand proves more cost-effective than on-site energy generation. ZEBs achieve self-sufficiency by integrating RE sources and advanced energy-efficient technologies. PEBs not only meet their own energy needs but also produce surplus energy, substantially contributing to enhanced energy efficiency and sustainability.

The trend towards reducing energy consumption and emissions from buildings is shifting towards a macro-level approach, specifically focusing on communities of buildings. A community of buildings consists of a group of buildings sharing common areas in proximity and can lead to advantages such as load diversity, the ability to control, store, and divert where the community produces more energy than it consumes, leading to enhanced energy resilience, cost savings, and a reduction in greenhouse gas emissions. This study thoroughly examines the energy efficiency of both individual buildings and communities, pinpointing key challenges and offering strategic solutions. The primary challenges include setting macro-level targets, accurately calculating energy demand and supply, enhancing the building envelope, and aligning various energy systems. These issues are systematically organized into four essential categories: building physics, HVAC and energy systems, the performance of individual buildings, and energy dynamics within communities. Additionally, we have presented a detailed comparison of our results with existing literature to highlight the advancements and unique contributions of our research. This comparison underscores the effectiveness of our proposed boundaries and frameworks in enhancing energy efficiency and sustainability in building areas.

or equal to the on-site renewable exported energy. The latest approach is to develop PECs,

**Boundary 1: Building Envelope:** Optimizing building design to reduce energy consumption involves several building physics strategies. These include natural light utilization, passive design techniques, proper insulation, energy-efficient windows, and ventilation systems. Additionally, using high-reflectance roofing materials, shading devices, and incorporating passive solar design elements can increase energy efficiency. Envelope component matching, where building envelope systems are configured based on annual energy consumption patterns, is essential for optimizing energy demand within a building.

**Boundary 2: HVAC and Energy Systems:** The selection, integration, and optimization of energy systems in buildings are complex processes influenced by various factors. Proper HVAC equipment sizing, zoning, and routine maintenance are crucial for reducing energy waste, improving efficiency, and enhancing thermal comfort. Energy storage systems are vital for balancing energy supply and demand. However, developing standard, cost-effective, and rational energy storage systems for heating, cooling, and electricity is necessary to achieve optimal performance. While integrating EVs into buildings holds potential, the high frequency of charging and discharging remains a significant barrier to their widespread adoption.

Boundary 3: Individual Buildings: The pursuit of nZEBs, ZEBs, and PEBs involves overcoming several challenges, including technical, financial, behavioral, regulatory, monitoring, and interdisciplinary issues. Technical challenges encompass designing and constructing energy-efficient building systems, integrating RE sources, optimizing building performance, and adhering to technical standards and regulations. Addressing these challenges necessitates an interdisciplinary approach. Financial challenges include elevated construction expenses, lack of financing, and a scarcity of incentives for sustainable building practices. Behavioral challenges relate to human attitudes and operations and maintenance practices. Consistent implementation and adherence to regulations and standards pose additional challenges, as do monitoring and verification processes. Overcoming these challenges requires collaboration among governments, building industry professionals, and educational institutions to establish regulations, promote transparency and accountability, and provide technical expertise for effective monitoring and verification of energy performance. Interdisciplinary challenges require effective communication, collaboration, and mutual understanding of roles and goals to create highly energy-efficient and sustainable buildings.

**Boundary 4: Communities of Buildings:** The pursuit of nZECs, ZECs, and PECs is increasingly important for reducing energy consumption and improving energy efficiency. The scale of a community of buildings significantly influences overall energy performance, necessitating the design of energy efficiency strategies that consider local climate conditions

for optimal results. Integrating energy efficiency into community building design and construction is critical for achieving nZECs, ZECs, and PECs. Advanced technologies, such as real-time monitoring and advanced data analytics, support the development of more effective and efficient community-level energy efficiency strategies. The method of allocating remaining grid imports among community buildings critically impacts energy performance, emphasizing the need to establish a fair and widely accepted approach for energy distribution within building communities. However, the lack of a clear definition and methodology for evaluating these communities and the challenges of coordinating energy-efficient design, construction, and operational practices across multiple buildings pose significant obstacles to their realization.

This research builds on and expands the existing body of knowledge by providing a comprehensive framework that integrates RE sources, energy storage systems, and real-time monitoring technologies. This holistic approach ensures optimal energy flow and equitable distribution within building communities. The proposed framework has demonstrated greater effectiveness in achieving energy efficiency compared to previously documented isolated strategies. Furthermore, our findings underscore the importance of considering local climate conditions and the specific energy needs of the community, aspects often overlooked in earlier studies. This comprehensive comparison highlights the practical application and significant impact of our proposed boundaries and strategies in enhancing energy resilience and sustainability in building communities.

## 6. Conclusions

This research provides an in-depth analysis of energy efficiency in both individual buildings and communities. It identifies key obstacles and proposes strategic solutions to address them. The main challenges involve establishing overarching targets, precisely determining energy demand and supply, improving building envelopes, and coordinating various energy systems. These challenges are categorized into four main areas: building physics, HVAC and energy systems, performance optimization of individual buildings, and energy interactions within communities. By addressing these areas, this study offers a comprehensive approach to enhancing energy efficiency, emphasizing the importance of integrated strategies and collaborative efforts among stakeholders. The insights gained from this research contribute significantly to the development of sustainable and resilient building practices, paving the way for future advancements in the field.

Effective collaboration among government agencies, industry professionals, and academic institutions is imperative to establish robust and consistent regulations and standards. Such cooperation will enhance transparency and accountability, ensuring the successful implementation of energy-efficient initiatives. Additionally, providing comprehensive technical training and expertise is essential to support these efforts. Integrating city planning with building design into a cohesive energy planning process is vital for achieving the ambitious goals of sustainable and resilient communities.

In summary, this research contributes to the existing body of knowledge by presenting a comprehensive framework that addresses the integration of RE sources, energy storage systems, and advanced monitoring technologies. The framework's effectiveness in optimizing energy flow and ensuring equitable energy distribution within building communities is a significant advancement over previous strategies. By considering local climate conditions and specific community energy needs, our approach offers a practical and impactful solution for enhancing energy efficiency and sustainability in the built environment.

**Author Contributions:** Conceptualization, B.V.; methodology, B.V.; software, X.C.; validation, X.C.; formal analysis, B.V.; investigation, X.C. and B.V.; resources, S.B.; data curation, X.C.; writing—original draft preparation, X.C., B.V. and S.B.; writing—review and editing, X.C., B.V. and S.B.; visualization, X.C.; supervision, B.V. and S.B.; project administration, B.V. and S.B.; funding acquisition, S.B. All authors have read and agreed to the published version of the manuscript.

27 of 32

**Funding:** The first and third authors were supported by the National Key R&D Program of China grant 2022YFE0198700, and by the Natural Science Foundation of China grants 62150610499 and 62073074.

**Data Availability Statement:** The raw data supporting the conclusions of this article will be made available by the authors on request.

Conflicts of Interest: The authors declare no conflicts of interest.

#### Abbreviations

The following abbreviations are used in this manuscript:

- BIPV Building-integrated Photovoltaic
- CHP Combined Heat and Power
- EMS Energy Management System
- EV Electric Vehicle
- GSHP Ground Source Heat Pump
- HVAC Heating, Ventilation and Air Conditioning
- NEB Negative Energy Building
- NEC Negative Energy Community
- nZEB Nearly Zero Energy Building
- nZEC Nearly Zero Energy Community
- PEB Positive Energy Building
- PEC Positive Energy Community
- PED Positive Energy District
- PV Photovoltaic
- RE Renewable Energy
- STC Solar Thermal Collector
- ZEB Zero Energy Building
- ZEC Zero Energy Community

## References

- 1. Singh, R.; Lazarus, I.J. Energy-Efficient Building Construction and Embodied Energy. In *Sustainability through Energy-Efficient Buildings*; CRC Press: Boca Raton, FL, USA, 2018.
- Mavrokapnidis, D.; Mitropoulou, C.C.; Lagaros, N.D. Environmental Assessment of Cost Optimized Structural Systems in Tall Buildings. J. Build. Eng. 2019, 24, 100730. [CrossRef]
- Hedman, Å.; Rehman, H.U.; Gabaldón, A.; Bisello, A.; Albert-Seifried, V.; Zhang, X.; Guarino, F.; Grynning, S.; Eicker, U.; Neumann, H.M.; et al. IEA EBC Annex83 Positive Energy Districts. *Buildings* 2021, 11, 130. [CrossRef]
- Xiang, X.; Ma, M.; Ma, X.; Chen, L.; Cai, W.; Feng, W.; Ma, Z. Historical Decarbonization of Global Commercial Building Operations in the 21st Century. *Appl. Energy* 2022, 322, 119401. [CrossRef]
- Sun, Y.; Wilson, R.; Wu, Y. A Review of Transparent Insulation Material (TIM) for Building Energy Saving and Daylight Comfort. *Appl. Energy* 2018, 226, 713–729. [CrossRef]
- 6. Yüksek, I.; Karadayi, T.T. Energy-Efficient Building Design in the Context of Building Life Cycle. In *Energy Efficient Buildings*; Yap, E.H., Ed.; InTech: London, UK, 2017. [CrossRef]
- 7. Zhou, Y.; Liu, Z. A Cross-Scale 'Material-Component-System' Framework for Transition towards Zero-Carbon Buildings and Districts with Low, Medium and High-Temperature Phase Change Materials. *Sustain. Cities Soc.* **2023**, *89*, 104378. [CrossRef]
- Technological Advancements towards the Net-Zero Energy Communities: A Review on 23 Case Studies around the Globe. Sol. Energy 2021, 224, 1107–1126. [CrossRef]
- Nalcaci, G.; Nalcaci, G. Modeling and Implementation of an Adaptive Facade Design for Energy Efficiently Buildings Based Biomimicry. In Proceedings of the 2020 8th International Conference on Smart Grid (icSmartGrid), Paris, France, 17–19 June 2020; pp. 140–145. [CrossRef]
- Economidou, M.; Todeschi, V.; Bertoldi, P.; D'Agostino, D.; Zangheri, P.; Castellazzi, L. Review of 50 years of EU Energy Efficiency Policies for Buildings. *Energy Build*. 2020, 225, 110322. [CrossRef]
- 11. Ruparathna, R.; Hewage, K.; Sadiq, R. Improving the Energy Efficiency of the Existing Building Stock: A Critical Review of Commercial and Institutional Buildings. *Renew. Sustain. Energy Rev.* **2016**, *53*, 1032–1045. [CrossRef]
- 12. Lee, D.Y.; Seo, B.M.; Yoon, Y.B.; Hong, S.H.; Choi, J.M.; Lee, K.H. Heating Energy Performance and Part Load Ratio Characteristics of Boiler Staging in an Office Building. *Front. Energy* **2019**, *13*, 339–353. [CrossRef]
- Baldi, S.; Michailidis, I.; Ravanis, C.; Kosmatopoulos, E.B. Model-Based and Model-Free "Plug-and-Play" Building Energy Efficient Control. *Appl. Energy* 2015, 154, 829–841. [CrossRef]

- Wang, Y.; Zhao, F.Y.; Kuckelkorn, J.; Li, X.H.; Wang, H.Q. Indoor Air Environment and Night Cooling Energy Efficiency of a Southern German Passive Public School Building Operated by the Heat Recovery Air Conditioning Unit. *Energy Build.* 2014, *81*, 9–17. [CrossRef]
- Liu, Z.; Zhou, Y.; Yan, J.; Tostado-Véliz, M. Frontier Ocean Thermal/Power and Solar PV Systems for Transformation towards Net-Zero Communities. *Energy* 2023, 284, 128362. [CrossRef]
- 16. Swaminathan, S.; Wang, X.; Zhou, B.; Baldi, S. A University Building Test Case for Occupancy-Based Building Automation. *Energies* **2018**, *11*, 3145. [CrossRef]
- REN21. Renewables 2021 Global Status Report. Paris. 2021. Available online: https://www.ren21.net/reports/global-status-report (accessed on 9 May 2024).
- Panwar, N.L.; Kaushik, S.C.; Kothari, S. Role of Renewable Energy Sources in Environmental Protection: A Review. *Renew. Sustain. Energy Rev.* 2011, 15, 1513–1524. [CrossRef]
- 19. Ordóñez, J.; Jadraque, E.; Alegre, J.; Martínez, G. Analysis of the Photovoltaic Solar Energy Capacity of Residential Rooftops in Andalusia (Spain). *Renew. Sustain. Energy Rev.* 2010, 14, 2122–2130. [CrossRef]
- 20. Kammen, D.M.; Sunter, D.A. City-Integrated Renewable Energy for Urban Sustainability. Science 2016, 352, 922–928. [CrossRef]
- 21. Korkas, C.D.; Baldi, S.; Michailidis, I.; Kosmatopoulos, E.B. Occupancy-Based Demand Response and Thermal Comfort Optimization in Microgrids with Renewable Energy Sources and Energy Storage. *Appl. Energy* **2016**, *163*, 93–104. [CrossRef]
- 22. A Framework for Analyzing City-Wide Impact of Building-Integrated Renewable Energy. *Appl. Energy* 2020, 276, 115489. [CrossRef]
- 23. A Review of Renewable Energy Assessment Methods in Green Building and Green Neighborhood Rating Systems. *Energy Build*. **2019**, *195*, 68–81. [CrossRef]
- 24. Jiang, Z.; Rahimi-Eichi, H. Design, Modeling and Simulation of a Green Building Energy System. In Proceedings of the 2009 IEEE Power & Energy Society General Meeting, Calgary, AB, Canada, 26–30 July 2009; pp. 1–7. [CrossRef]
- Zalamea-León, E.; Mena-Campos, J.; Barragán-Escandón, A.; Parra-González, D.; Méndez-Santos, P. Urban Photovoltaic Potential of Inclined Roofing for Buildings in Heritage Centers in Equatorial Areas. J. Green Build. 2018, 13, 45–69. [CrossRef]
- 26. Menegazzo, D.; Lombardo, G.; Bobbo, S.; De Carli, M.; Fedele, L. State of the Art, Perspective and Obstacles of Ground-Source Heat Pump Technology in the European Building Sector: A Review. *Energies* **2022**, *15*, 2685. [CrossRef]
- Gunkel, P.A.; Bergaentzlé, C.; Jensen, I.G.; Scheller, F. From Passive to Active: Flexibility from Electric Vehicles in the Context of Transmission System Development. *Appl. Energy* 2020, 277, 115526. [CrossRef]
- Omrany, H.; Chang, R.; Soebarto, V.; Zhang, Y.; Ghaffarianhoseini, A.; Zuo, J. A Bibliometric Review of Net Zero Energy Building Research 1995–2022. Energy Build. 2022, 262, 111996. [CrossRef]
- Guo, D.; Zhou, C. Potential Performance Analysis and Future Trend Prediction of Electric Vehicle with V2G/V2H/V2B Capability. AIMS Energy 2016, 4, 331–346. [CrossRef]
- 30. Bishop, J.D.K.; Axon, C.J.; Bonilla, D.; Tran, M.; Banister, D.; McCulloch, M.D. Evaluating the Impact of V2G Services on the Degradation of Batteries in PHEV and EV. *Appl. Energy* **2013**, *111*, 206–218. [CrossRef]
- Korkas, C.D.; Baldi, S.; Kosmatopoulos, E.B. 9-Grid-Connected Microgrids: Demand Management via Distributed Control and Human-in-the-Loop Optimization. In *Advances in Renewable Energies and Power Technologies*; Yahyaoui, I., Ed.; Elsevier: Amsterdam, The Netherlands, 2018; pp. 315–344. [CrossRef]
- Zhou, Y.; Cao, S.; Hensen, J.L.M.; Hasan, A. Heuristic Battery-Protective Strategy for Energy Management of an Interactive Renewables–Buildings–Vehicles Energy Sharing Network with High Energy Flexibility. *Energy Convers. Manag.* 2020, 214, 112891. [CrossRef]
- Barone, G.; Buonomano, A.; Forzano, C.; Giuzio, G.F.; Palombo, A.; Russo, G. Energy Virtual Networks Based on Electric Vehicles for Sustainable Buildings: System Modelling for Comparative Energy and Economic Analyses. *Energy* 2022, 242, 122931. [CrossRef]
- 34. Huda, M.; Koji, T.; Aziz, M. Techno Economic Analysis of Vehicle to Grid (V2G) Integration as Distributed Energy Resources in Indonesia Power System. *Energies* **2020**, *13*, 1162. [CrossRef]
- 35. United Nations Environment Programme. 2020 Global Status Report for Buildings and Construction: Towards a Zero-Emission, Efficient and Resilient Buildings and Construction Sector; United Nations Environment Programme: Nairobi, Kenya, 2020. Available online: https://globalabc.org/sites/default/files/2021-10/GABC\_Buildings-GSR-2021\_BOOK.pdf (accessed on 9 May 2024).
- 36. Wilberforce, T.; Olabi, A.G.; Sayed, E.T.; Elsaid, K.; Maghrabie, H.M.; Abdelkareem, M.A. A Review on Zero Energy Buildings Pros and Cons. *Energy Built Environ*. **2021**, *4*, 25–38. [CrossRef]
- Torcellini, P.; Pless, S.; Deru, M.; Crawley, D. Zero Energy Buildings: A Critical Look at the Definition; National Renewable Energy Laboratory (NREL): Golden, CO, USA, 2006.
- Cao, X.; Dai, X.; Liu, J. Building Energy-Consumption Status Worldwide and the State-of-the-Art Technologies for Zero-Energy Buildings during the Past Decade. *Energy Build*. 2016, 128, 198–213. [CrossRef]
- Magrini, A.; Lentini, G.; Cuman, S.; Bodrato, A.; Marenco, L. From Nearly Zero Energy Buildings (NZEB) to Positive Energy Buildings (PEB): The next Challenge - The Most Recent European Trends with Some Notes on the Energy Analysis of a Forerunner PEB Example. *Dev. Built Environ.* 2020, *3*, 100019. [CrossRef]
- 40. ur Rehman, H.; Hasan, A.; Reda, F. Challenges in Reaching Positive Energy Building Level in Apartment Buildings in the Nordic Climate: A Techno-Economic Analysis. *Energy Build.* 2022, 262, 111991. [CrossRef]

- 41. Kayo, G.; Hasan, A.; Siren, K. Energy Sharing and Matching in Different Combinations of Buildings, CHP Capacities and Operation Strategy. *Energy Build*. 2014, 82, 685–695. [CrossRef]
- 42. Vand, B.; Ruusu, R.; Hasan, A.; Manrique Delgado, B. Optimal Management of Energy Sharing in a Community of Buildings Using a Model Predictive Control. *Energy Convers. Manag.* 2021, 239, 114178. [CrossRef]
- Carlisle, N.; Van Geet, O.; Pless, S. *Definition of a 'Zero Net Energy' Community*; Technical Report NREL/TP-7A2-46065, 969716; National Renewable Energy Laboratory (NREL): Golden, CO, USA, 2009. [CrossRef]
- Sornes, K.; Sartori, I.; Fredriksen, E.; Martinsson, F.; Romero, A.; Rodriguez, F.; Schneuwly, P. ZenN Nearly Zero Energy Neighborhoods. Final Report on Common Definition for nZEB Renovation. 2014. Available online: https://hdl.handle.net/1125 0/2406471 (accessed on 9 May 2024).
- 45. Bremvåg, A.; Gustavsen, A.; Hestnes, A.G. ZEN Research Centre on Zero Emission Neighbourhoods in Smart Cities: Annual Report 2017. 2018. Available online: https://sintef.brage.unit.no/sintef-xmlui/handle/11250/2588746 (accessed on 9 May 2024).
- Brozovsky, J.; Gustavsen, A.; Gaitani, N. Zero Emission Neighbourhoods and Positive Energy Districts A State-of-the-Art Review. Sustain. Cities Soc. 2021, 72, 103013. [CrossRef]
- SET-Plan ACTION n°3.2 Implementation Plan. Europe to Become a Global Role Model in Integrated, Innovative Solutions for the Planning, Deployment, and Replication of Positive Energy Districts. Available online: <a href="https://jpi-urbaneurope.eu/wp-content/uploads/2021/10/setplan\_smartcities\_implementationplan-2.pdf">https://jpi-urbaneurope.eu/wp-content/uploads/2021/10/setplan\_smartcities\_implementationplan-2.pdf</a> (accessed on 9 May 2024).
- Čaušević, S.; Huitema, G.B.; Subramanian, A.; van Leeuwen, C.; Konsman, M. Towards Positive Energy Districts in Smart Cities: A Data-Driven Approach Using Aggregation and Disaggregation of Energy Balance Calculations. *Environ. Sci. Proc.* 2021, 11, 1. [CrossRef]
- 49. Lindholm, O.; ur Rehman, H.; Reda, F. Positioning Positive Energy Districts in European Cities. Buildings 2021, 11, 19. [CrossRef]
- 50. Mihailova, D.; Schubert, I.; Martinez-Cruz, A.L.; Hearn, A.X.; Sohre, A. Preferences for Configurations of Positive Energy Districts – Insights from a Discrete Choice Experiment on Swiss Households. *Energy Policy* **2022**, *163*, 112824. [CrossRef]
- 51. White Paper on Reference Framework for Positive Energy Districts and Neighbourhoods. Available online: https://policycommons.net/artifacts/2053524/white-paper/2806615/ (accessed on 9 May 2024).
- Clerici Maestosi, P.; Andreucci, M.B.; Civiero, P. Sustainable Urban Areas for 2030 in a Post-COVID-19 Scenario: Focus on Innovative Research and Funding Frameworks to Boost Transition towards 100 Positive Energy Districts and 100 Climate-Neutral Cities. *Energies* 2021, 14, 216. [CrossRef]
- 53. Zohuri, B. First Law of Thermodynamics. In *Physics of Cryogenics*; Elsevier: Amsterdam, The Netherlands, 2018; pp. 119–163. [CrossRef]
- 54. Sarihi, S.; Mehdizadeh Saradj, F.; Faizi, M. A Critical Review of Façade Retrofit Measures for Minimizing Heating and Cooling Demand in Existing Buildings. *Sustain. Cities Soc.* **2021**, *64*, 102525. [CrossRef]
- 55. Hashempour, N.; Taherkhani, R.; Mahdikhani, M. Energy Performance Optimization of Existing Buildings: A Literature Review. *Sustain. Cities Soc.* 2020, 54, 101967. [CrossRef]
- Marino, C.; Minichiello, F. Existing Buildings and HVAC Systems: Incidence of Innovative Surface Finishes on the Energy Requirements. *Energy Procedia* 2015, 82, 499–505. [CrossRef]
- Gou, Z.; Lau, S.S.Y. Contextualizing Green Building Rating Systems: Case Study of Hong Kong. *Habitat Int.* 2014, 44, 282–289. [CrossRef] [PubMed]
- 58. Ma, Z.; Cooper, P.; Daly, D.; Ledo, L. Existing Building Retrofits: Methodology and State-of-the-Art. *Energy Build*. 2012, 55, 889–902. [CrossRef]
- Sun, X.; Gou, Z.; Lau, S.S.Y. Cost-Effectiveness of Active and Passive Design Strategies for Existing Building Retrofits in Tropical Climate: Case Study of a Zero Energy Building. J. Clean. Prod. 2018, 183, 35–45. [CrossRef]
- 60. Wang, Z.; Ding, Y.; Geng, G.; Zhu, N. Analysis of Energy Efficiency Retrofit Schemes for Heating, Ventilating and Air-Conditioning Systems in Existing Office Buildings Based on the Modified Bin Method. *Energy Convers. Manag.* **2014**, *77*, 233–242. [CrossRef]
- Directive 2010/31/EU of the European Parliament and of the Council of 19 May 2010 on the Energy Performance of Buildings. Available online: https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2010:153:0013:0035:EN:PDF (accessed on 9 May 2024).
- 62. Liu, Z.; Zhou, Q.; Tian, Z.; He, B.j.; Jin, G. A Comprehensive Analysis on Definitions, Development, and Policies of Nearly Zero Energy Buildings in China. *Renew. Sustain. Energy Rev.* **2019**, *114*, 109314. [CrossRef]
- Feng, W.; Zhang, Q.; Ji, H.; Wang, R.; Zhou, N.; Ye, Q.; Hao, B.; Li, Y.; Luo, D.; Lau, S.S.Y. A Review of Net Zero Energy Buildings in Hot and Humid Climates: Experience Learned from 34 Case Study Buildings. *Renew. Sustain. Energy Rev.* 2019, 114, 109303. [CrossRef]
- 64. Amani, N.; Kiaee, E. Developing a Two-Criteria Framework to Rank Thermal Insulation Materials in Nearly Zero Energy Buildings Using Multi-Objective Optimization Approach. *J. Clean. Prod.* **2020**, *276*, 122592. [CrossRef]
- 65. Ferrari, S.; Beccali, M. Energy-Environmental and Cost Assessment of a Set of Strategies for Retrofitting a Public Building toward Nearly Zero-Energy Building Target. *Sustain. Cities Soc.* **2017**, *32*, 226–234. [CrossRef]
- 66. Hamdy, M.; Hasan, A.; Siren, K. A Multi-Stage Optimization Method for Cost-Optimal and Nearly-Zero-Energy Building Solutions in Line with the EPBD-recast 2010. *Energy Build*. 2013, *56*, 189–203. [CrossRef]
- 67. Zhao, L.; Zhang, H.; Wang, Q.; Wang, H. Digital-Twin-Based Evaluation of Nearly Zero-Energy Building for Existing Buildings Based on Scan-to-BIM. *Adv. Civ. Eng.* **2021**, 2021, e6638897. [CrossRef]

- 68. Visa, I.; Moldovan, M.D.; Comsit, M.; Duta, A. Improving the Renewable Energy Mix in a Building toward the Nearly Zero Energy Status. *Energy Build*. 2014, *68*, 72–78. [CrossRef]
- Panagiotidou, M.; Fuller, R.J. Progress in ZEBs—A Review of Definitions, Policies and Construction Activity. *Energy Policy* 2013, 62, 196–206. [CrossRef]
- Laustsen, J. Energy Efficiency Requirements in Building Codes, Energy Efficiency Policies for New Buildings. IEA Information Paper 2008. Available online: https://www.osti.gov/etdeweb/biblio/971038 (accessed on 9 May 2024).
- 71. MEPs Back Plans for a Climate Neutral Building Sector by 2050. 2023. Available online: https://www.europarl.europa.eu/news/
- en/press-room/20230310IPR77228/meps-back-plans-for-a-climate-neutral-building-sector-by-2050 (accessed on 9 May 2024).
- 72. Kylili, A.; Fokaides, P.A. European Smart Cities: The Role of Zero Energy Buildings. Sustain. Cities Soc. 2015, 15, 86–95. [CrossRef]
- 73. Kwan, Y.; Guan, L. Design a Zero Energy House in Brisbane, Australia. Procedia Eng. 2015, 121, 604–611. [CrossRef]
- Rey-Hernández, J.M.; Yousif, C.; Gatt, D.; Velasco-Gómez, E.; San José-Alonso, J.; Rey-Martínez, F.J. Modelling the Long-Term Effect of Climate Change on a Zero Energy and Carbon Dioxide Building through Energy Efficiency and Renewables. *Energy Build.* 2018, 174, 85–96. [CrossRef]
- Nord, N.; Qvistgaard, L.H.; Cao, G. Identifying Key Design Parameters of the Integrated Energy System for a Residential Zero Emission Building in Norway. *Renew. Energy* 2016, 87, 1076–1087. [CrossRef]
- Lou, S.; Tsang, E.K.W.; Li, D.H.W.; Lee, E.W.M.; Lam, J.C. Towards Zero Energy School Building Designs in Hong Kong. *Energy Procedia* 2017, 105, 182–187. [CrossRef]
- 77. Lindberg, K.B.; Fischer, D.; Doorman, G.; Korpås, M.; Sartori, I. Cost-Optimal Energy System Design in Zero Energy Buildings with Resulting Grid Impact: A Case Study of a German Multi-Family House. *Energy Build.* **2016**, *127*, 830–845. [CrossRef]
- Zomer, C.; Custódio, I.; Antoniolli, A.; Rüther, R. Performance Assessment of Partially Shaded Building-Integrated Photovoltaic (BIPV) Systems in a Positive-Energy Solar Energy Laboratory Building: Architecture Perspectives. Sol. Energy 2020, 211, 879–896. [CrossRef]
- 79. Kumar, G.M.S.; Cao, S. State-of-the-Art Review of Positive Energy Building and Community Systems. *Energies* **2021**, *14*, 5046. [CrossRef]
- 80. Cole, R.J.; Fedoruk, L. Shifting from Net-Zero to Net-Positive Energy Buildings. Build. Res. Inf. 2015, 43, 111–120. [CrossRef]
- 81. Ala-Juusela, M.; ur Rehman, H.; Hukkalainen, M.; Reda, F. Positive Energy Building Definition with the Framework, Elements and Challenges of the Concept. *Energies* 2021, 14, 6260. [CrossRef]
- 82. Ai, B.; Yang, H.; Shen, H.; Liao, X. Computer-Aided Design of PV/Wind Hybrid System. *Renew. Energy* 2003, 28, 1491–1512. [CrossRef]
- 83. Kolokotsa, D.; Vagias, V.; Fytraki, L.; Oungrinis, K. Energy Analysis of Zero Energy Schools: The Case Study of Child's Asylum in Greece. *Adv. Build. Energy Res.* 2019, *13*, 193–204. [CrossRef]
- Bennani, O.; Bensaadout, I.; Ouassaid, M. Positive Energy Office Building: A Case Study in Casablanca, Morocco. In Proceedings of the 2016 International Renewable and Sustainable Energy Conference (IRSEC), Marrakech, Morocco, 14–17 November 2016; pp. 775–780. [CrossRef]
- Casini, M. A Positive Energy Building for the Middle East Climate: ReStart4Smart Solar House at Solar Decathlon Middle East 2018. *Renew. Energy* 2020, 159, 1269–1296. [CrossRef]
- Zomer, C.; Custódio, I.; Goulart, S.; Mantelli, S.; Martins, G.; Campos, R.; Pinto, G.; Rüther, R. Energy Balance and Performance Assessment of PV Systems Installed at a Positive-Energy Building (PEB) Solar Energy Research Centre. *Sol. Energy* 2020, 212, 258–274. [CrossRef]
- Mavrigiannaki, A.; Pignatta, G.; Assimakopoulos, M.; Isaac, M.; Gupta, R.; Kolokotsa, D.; Laskari, M.; Saliari, M.; Meir, I.A.; Isaac, S. Examining the Benefits and Barriers for the Implementation of Net Zero Energy Settlements. *Energy Build.* 2021, 230, 110564. [CrossRef]
- 88. Kopanos, G.; Liu, P.; Georgiadis, M. *Advances in Energy Systems Engineering*; Springer: Cham, Switzerland, 2017. https://linkspringer.53yu.com/book/10.1007/978-3-319-42803-1.
- ur Rehman, H.; Reda, F.; Paiho, S.; Hasan, A. Towards Positive Energy Communities at High Latitudes. *Energy Convers. Manag.* 2019, 196, 175–195. [CrossRef]
- 90. Foley, A.M.; Leahy, P.G.; Marvuglia, A.; McKeogh, E.J. Current Methods and Advances in Forecasting of Wind Power Generation. *Renew. Energy* **2012**, *37*, 1–8. [CrossRef]
- Mittal, A.; Krejci, C.C.; Dorneich, M.C.; Fickes, D. An Agent-Based Approach to Modeling Zero Energy Communities. Sol. Energy 2019, 191, 193–204. [CrossRef]
- 92. Charani Shandiz, S.; Rismanchi, B.; Foliente, G. Energy Master Planning for Net-Zero Emission Communities: State of the Art and Research Challenges. *Renew. Sustain. Energy Rev.* 2021, 137, 110600. [CrossRef]
- 93. Huang, S.; Wang, J.; Fu, Y.; Zuo, W.; Hinkelman, K.; Kaiser, R.M.; He, D.; Vrabie, D. An Open-Source Virtual Testbed for a Real Net-Zero Energy Community. *Sustain. Cities Soc.* 2021, 75, 103255. [CrossRef]
- Prasad, A.; Dusparic, I. Multi-Agent Deep Reinforcement Learning for Zero Energy Communities. In Proceedings of the 2019 IEEE PES Innovative Smart Grid Technologies Europe (ISGT-Europe), Bucharest, Romania, 29 September 2019–2 October 2019; pp. 1–5. [CrossRef]
- 95. Barbosa, J.A.S.D.; Araújo, C.; Bragança, L.; Mateus, R. Study of the of the Concept of Community Buildings and Its Importance for Land Use Efficiency. In Proceedings of the Euro-ELECS, Guimaraes, Portugal, 21–23 July 2015. [CrossRef]

- 96. Ghofrani, A.; Nazemi, S.D.; Jafari, M.A. HVAC Load Synchronization in Smart Building Communities. *Sustain. Cities Soc.* 2019, 51, 101741. [CrossRef]
- Amaral, A.R.; Rodrigues, E.; Rodrigues Gaspar, A.; Gomes, Á. Review on Performance Aspects of Nearly Zero-Energy Districts. Sustain. Cities Soc. 2018, 43, 406–420. [CrossRef]
- Synnefa, A.; Laskari, M.; Gupta, R.; Pisello, A.L.; Santamouris, M. Development of Net Zero Energy Settlements Using Advanced Energy Technologies. *Procedia Eng.* 2017, 180, 1388–1401. [CrossRef]
- 99. Hachem-Vermette, C.; Cubi, E.; Bergerson, J. Energy Performance of a Solar Mixed-Use Community. *Sustain. Cities Soc.* 2016, 27, 145–151. [CrossRef]
- Sameti, M.; Haghighat, F. Integration of Distributed Energy Storage into Net-Zero Energy District Systems: Optimum Design and Operation. *Energy* 2018, 153, 575–591. [CrossRef]
- Wills, A.D.; Beausoleil-Morrison, I.; Ugursal, V.I. A Modelling Approach and a Case Study to Answer the Question: What Does It Take to Retrofit a Community to Net-Zero Energy? *J. Build. Eng.* 2021, 40, 102296. [CrossRef]
- Katipamula, S.; Fernandez, N.; Brambley, M.R.; Reddy, T.A. Building-Scale vs. Community-Scale Net-Zero Energy Performance. Available online: https://aceee.org/files/proceedings/2010/data/papers/1951.pdf (accessed on 9 May 2024).
- 103. Mavrigiannaki, A.; Gobakis, K.; Kolokotsa, D.; Kalaitzakis, K.; Pisello, A.L.; Piselli, C.; Laskari, M.; Saliari, M.; Assimakopoulos, M.N.; Pignatta, G.; et al. Zero Energy Concept at Neighborhood Level: A Case Study Analysis. *Sol. Energy Adv.* 2021, 1, 100002. [CrossRef]
- Fouad, M.M.; Iskander, J.; Shihata, L.A. Energy, Carbon and Cost Analysis for an Innovative Zero Energy Community Design. Sol. Energy 2020, 206, 245–255. [CrossRef]
- European Commission. Joint Research Centre. Enabling Positive Energy Districts across Europe: Energy Efficiency Couples Renewable Energy; Publications Office: Luxembourg, 2020. Available online: https://data.europa.eu/doi/10.2760/452028 (accessed on 9 May 2024).
- 106. Morales González, R.; Asare-Bediako, B.; Cobben, J.; Kling, W.; Scharrenberg, G.; Dijkstra, D. Distributed Energy Resources for a Zero-Energy Neighborhood. In Proceedings of the 2012 3rd IEEE PES Innovative Smart Grid Technologies Europe (ISGT Europe), Berlin, Germany, 14–17 October 2012; pp. 1–8. [CrossRef]
- 107. De Leon, L.; Cedeño, M.; Mora, D.; Chen Austin, M. Towards A Definition For Zero Energy Districts In Panama: A Numerical Assessment Of Passive And Active Strategies. In Proceedings of the 19th LACCEI International Multi-Conference for Engineering, Education, and Technology: "Prospective and Trends in Technology and Skills for Sustainable Social Development" "Leveraging Emerging Technologies to Construct the Future", Virtual, 19–23 July 2021. [CrossRef]
- Derkenbaeva, E.; Halleck Vega, S.; Hofstede, G.J.; van Leeuwen, E. Positive Energy Districts: Mainstreaming Energy Transition in Urban Areas. *Renew. Sustain. Energy Rev.* 2022, 153, 111782. [CrossRef]
- 109. Tuerk, A.; Frieden, D.; Neumann, C.; Latanis, K.; Tsitsanis, A.; Kousouris, S.; Llorente, J.; Heimonen, I.; Reda, F.; Ala-Juusela, M.; et al. Integrating Plus Energy Buildings and Districts with the EU Energy Community Framework: Regulatory Opportunities, Barriers and Technological Solutions. *Buildings* 2021, *11*, 468. [CrossRef]
- 110. Salom, J.; Tamm, M.; Andresen, I.; Cali, D.; Magyari, Á.; Bukovszki, V.; Balázs, R.; Dorizas, P.V.; Toth, Z.; Zuhaib, S.; et al. An Evaluation Framework for Sustainable Plus Energy Neighbourhoods: Moving Beyond the Traditional Building Energy Assessment. *Energies* 2021, 14, 4314. [CrossRef]
- 111. Uspenskaia, D.; Specht, K.; Kondziella, H.; Bruckner, T. Challenges and Barriers for Net-Zero/Positive Energy Buildings and Districts—Empirical Evidence from the Smart City Project SPARCS. *Buildings* **2021**, *11*, 78. [CrossRef]
- 112. Barone, G.; Buonomano, A.; Forzano, C.; Giuzio, G.F.; Palombo, A. Increasing Self-Consumption of Renewable Energy through the Building to Vehicle to Building Approach Applied to Multiple Users Connected in a Virtual Micro-Grid. *Renew. Energy* **2020**, 159, 1165–1176. [CrossRef]
- Brennenstuhl, M.; Zeh, R.; Otto, R.; Pesch, R.; Stockinger, V.; Pietruschka, D. Report on a Plus-Energy District with Low-Temperature DHC Network, Novel Agrothermal Heat Source, and Applied Demand Response. *Appl. Sci.* 2019, *9*, 5059. [CrossRef]
- 114. Bambara, J.; Athienitis, A.K.; Eicker, U. Residential Densification for Positive Energy Districts. *Front. Sustain. Cities* **2021**, *3*, 630973. [CrossRef]
- 115. Gouveia, J.P.; Seixas, J.; Palma, P.; Duarte, H.; Luz, H.; Cavadini, G.B. Positive Energy District: A Model for Historic Districts to Address Energy Poverty. *Front. Sustain. Cities* **2021**, *3*, 648473. [CrossRef]
- 116. European Commission, FAQ-Work Programme 2018 for Horizon 2020–Smart Cities and Communities–Lighthouse Projects. Available online: https://ec.europa.eu/research/participants/portal4/doc/call/h2020/lc-sc3-scc-1-2018-2019-2020/1800106 -faq\_ssc1\_2018\_v07\_en.pdf (accessed on 9 May 2024).
- Gernaat, D.E.H.J.; de Boer, H.S.; Daioglou, V.; Yalew, S.G.; Müller, C.; van Vuuren, D.P. Climate Change Impacts on Renewable Energy Supply. *Nat. Clim. Chang.* 2021, 11, 119–125. [CrossRef]
- 118. Chen, Q.; Kuang, Z.; Liu, X.; Zhang, T. Transforming a Solar-Rich County to an Electricity Producer: Solutions to the Mismatch between Demand and Generation. *J. Clean. Prod.* **2022**, *336*, 130418. [CrossRef]
- Chen, X.; Xiao, J.; Yuan, J.; Xiao, Z.; Gang, W. Application and Performance Analysis of 100% Renewable Energy Systems Serving Low-Density Communities. *Renew. Energy* 2021, 176, 433–446. [CrossRef]

- Gabaldón Moreno, A.; Vélez, F.; Alpagut, B.; Hernández, P.; Sanz Montalvillo, C. How to Achieve Positive Energy Districts for Sustainable Cities: A Proposed Calculation Methodology. *Sustainability* 2021, 13, 710. [CrossRef]
- Baetens, R.; De Coninck, R.; Van Roy, J.; Verbruggen, B.; Driesen, J.; Helsen, L.; Saelens, D. Assessing Electrical Bottlenecks at Feeder Level for Residential Net Zero-Energy Buildings by Integrated System Simulation. *Appl. Energy* 2012, 96, 74–83. [CrossRef]
- 122. Committees, A.T. 2021 ASHRAE Handbook—Fundamentals; American Society of Heating, Refrigeration, and Air-Conditioning Engineers (ASHRAE): Peachtree Corners, GA, USA, 2021. Available online: https://www.ashrae.org/technical-resources/ashrae-handbook (accessed on 9 May 2024).
- 123. United States Department of Energy, Office of Energy Efficiency & Renewable Energy, Tips on Saving Money and Energy in Your Home. 2022. Available online: https://www.energy.gov/energysaver/energy-saver-guide-tips-saving-money-and-energyhome (accessed on 9 May 2024).
- Cebulla, F.; Haas, J.; Eichman, J.; Nowak, W.; Mancarella, P. How Much Electrical Energy Storage Do We Need? A Synthesis for the U.S., Europe, and Germany. J. Clean. Prod. 2018, 181, 449–459. [CrossRef]
- Ohene, E.; Chan, A.P.C.; Darko, A. Review of Global Research Advances towards Net-Zero Emissions Buildings. *Energy Build*. 2022, 266, 112142. [CrossRef]
- 126. UNE-CEN/TR 16798-2:2019; Energy Performance of Buildings—Ventilation for Buildings—Part 2: Interpretation of the Requirements in EN 16798-1—Indoor Environmental Input Parameters for Design and Assessment of Energy Performance of Buildings Addressing Indoor Air Quality, Thermal Environment, Lighting and Acoustics (ModuleM1-6). UNE: Armidale, Australia, 2019. Available online: https://standards.iteh.ai/catalog/standards/cen/554e05d6-df03-4e67-8907-52a8dbd298f5/cen-tr-16798-2-2019 (accessed on 9 May 2024).
- 127. Alfano, F.R.D.; Pepe, D.; Riccio, G.; Vio, M.; Palella, B.I. On the Effects of the Mean Radiant Temperature Evaluation in the Assessment of Thermal Comfort by Dynamic Energy Simulation Tools. *Build. Environ.* **2023**, 236, 110254. [CrossRef]
- 128. Alfano, F.R.D.; Olesen, B.W.; Pepe, D.; Palella, B.I. Working with Different Building Energy Performance Tools: From Input Data to Energy and Indoor Temperature Predictions. *Energies* **2023**, *16*, 743. [CrossRef]
- 129. Van Den Brom, P.; Meijer, A.; Visscher, H. Performance Gaps in Energy Consumption: Household Groups and Building Characteristics. *Build. Res. Inf.* 2018, 46, 54–70. [CrossRef]
- Rafique, M.M.; Rehman, S.; Alhems, L.M. Developing Zero Energy and Sustainable Villages A Case Study for Communities of the Future. *Renew. Energy* 2018, 127, 565–574. [CrossRef]
- Liu, Z.; Yu, C.; Qian, Q.K.; Huang, R.; You, K.; Visscher, H.; Zhang, G. Incentive Initiatives on Energy-Efficient Renovation of Existing Buildings towards Carbon–Neutral Blueprints in China: Advancements, Challenges and Prospects. *Energy Build*. 2023, 296, 113343. [CrossRef]
- Liu, Z.; Zhang, X.; Sun, Y.; Zhou, Y. Advanced Controls on Energy Reliability, Flexibility and Occupant-Centric Control for Smart and Energy-Efficient Buildings. *Energy Build.* 2023, 297, 113436. [CrossRef]
- Zhang, Y.; Vand, B.; Baldi, S. A Review of Mathematical Models of Building Physics and Energy Technologies for Environmentally Friendly Integrated Energy Management Systems. *Buildings* 2022, 12, 238. [CrossRef]
- Norbu, S.; Couraud, B.; Robu, V.; Andoni, M.; Flynn, D. Modelling the Redistribution of Benefits from Joint Investments in Community Energy Projects. *Appl. Energy* 2021, 287, 116575. [CrossRef]
- Kataray, T.; Nitesh, B.; Yarram, B.; Sinha, S.; Cuce, E.; Shaik, S.; Vigneshwaran, P.; Roy, A. Integration of Smart Grid with Renewable Energy Sources: Opportunities and Challenges – A Comprehensive Review. *Sustain. Energy Technol. Assess.* 2023, 58, 103363. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.